

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Lergetporer, Philipp; Wößmann, Ludger

Conference Paper Income Contingency and the Electorate's Support for Tuition

Beiträge zur Jahrestagung des Vereins für Socialpolitik 2024: Upcoming Labor Market Challenges

Provided in Cooperation with:

Verein für Socialpolitik / German Economic Association

Suggested Citation: Lergetporer, Philipp; Wößmann, Ludger (2024) : Income Contingency and the Electorate's Support for Tuition, Beiträge zur Jahrestagung des Vereins für Socialpolitik 2024: Upcoming Labor Market Challenges, ZBW - Leibniz Information Centre for Economics, Kiel, Hamburg

This Version is available at: https://hdl.handle.net/10419/302330

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Income Contingency and the Electorate's Support for Tuition^{*}

Philipp Lergetporer and Ludger Woessmann[†]

Abstract

We show that the electorate's preferences for using tuition to finance higher education strongly depend on the design of the payment scheme. In representative surveys of the German electorate (N>25,000), experimentally replacing regular upfront by deferred income-contingent payments increases public support for tuition by 18 percentage points. The treatment turns a minority of 44 percent favoring tuition into a strong majority of 62 percent. Additional experiments reveal that the treatment effect similarly shows when framed as loan repayments, when increasing perceived political consequentiality, and in a survey of adolescents. Reduced fairness concerns and improved student situations act as strong mediators. While payment deferral and income threshold both contribute to the overall effect, the former element accounts for most of it.

Keywords: tuition, higher education finance, income-contingent loans, voting

JEL classification: H52, I22, D72

January 2, 2024

^{*} Together with what is now a separate paper (Lergetporer and Woessmann, 2023), a previous version of this paper circulated under the title "The Political Economy of Higher Education Finance: How Information and Design Affect Public Preferences for Tuition." For valuable comments and discussion, we would like to thank the editor Juanna Schrøter Joensen, three anonymous referees, Nick Barr, Eric Bettinger, Leonardo Bursztyn, Bruce Chapman, Sue Dynarski, Achim Goerres, Rick Hanushek, Wilbert van der Klaauw, Volker Meier, Richard Murphy, Ricardo Perez-Truglia, Randall Reback, Lisa Simon, Marty West, and seminar participants at Harvard, Stanford, Ohio, Prague, Munich, Konstanz, Freiburg, Karlsruhe, Nuremberg, Brasilia, ifo Institute, the FRBNY/NYU education seminar, the CESifo education meeting, the IZA workshop on education, interventions and experiments, and the annual meetings of SOLE in Raleigh, EEA in Mannheim, EALE online, the German Economic Association in Augsburg, and its economics of education group in Bamberg. We are grateful to Vera Freundl, Elisabeth Grewenig, Sarah Kersten, Laura Oestreich, Franziska Pfaehler, and Katharina Werner for their collaboration in designing and executing the surveys. This study is registered in the AEA RCT Registry as AEARCTR-0003873. Financial support by the Leibniz Competition (SAW-2014-ifo-2) and the German Science Foundation (CRC TRR 190) is gratefully acknowledged.

[†] Lergetporer: Technical University of Munich and ifo Institute; CESifo; philipp.lergetporer@tum.de. Woessmann: University of Munich and ifo Institute; Hoover Institution, Stanford University; CESifo, IZA, and CAGE; woessmann@ifo.de.

1. Introduction

A high level of human capital is essential for economic prosperity and the sustainability of public finances (e.g., Hanushek and Woessmann, 2008; Stantcheva, 2017). Therefore, policymakers' choices among alternative financing systems for higher education, which have differing implications for equitable and efficient access, carry particular policy importance. Deferred income-contingent payment schemes are an alternative to regular upfront tuition in financing higher education. In the income-contingent design, payments are due only after graduation and must be paid only if income exceeds a certain threshold. Economists have repeatedly argued that income-contingent financing schemes – which are in place, e.g., in Australia, England, and New Zealand – have attractive efficiency and equity properties (e.g., Friedman, 1955; Barr, 2004; Chapman, 2006; Jacobs and van der Ploeg, 2006; Chapman, Higgins, and Stiglitz, 2014; Dynarski, 2014; Diris and Ooghe, 2018; Barr et al., 2019). In particular, optimal dynamic tax theory shows that income-contingent repayment schemes can lead the joint system of education finance and taxation to optimality by providing incentives for human capital accumulation while insuring against income risk (Findeisen and Sachs, 2016; Stantcheva, 2017).

Whether policymakers adopt income-contingent tuition in the real world, however, depends on the political-economy question whether voters support it. The question of student financing of higher education features prominently in the political debates in many countries, including the two most recent presidential campaigns in the United States.¹ In Germany, all state governments that had introduced upfront tuition in the late 2000s lost power in the next election where the tuition topic featured highly, and the new governments immediately acted to abandon tuition.² The implementation of various tuition designs therefore depends not only on *economic* considerations – which have been studied closely – but especially on the *political* consequences in terms of electoral support, causal evidence on which is largely lacking. By alleviating fairness concerns and restrictions on student choices, the income-contingent design may raise voters' preferences for charging tuition compared to the more standard upfront design.

¹ For a recent example of public debate about loan costs in the income-contingent system in England, see The Guardian, May 22, 2022, https://www.theguardian.com/money/2022/may/22/fear-of-12-interest-on-student-loans-will-put-many-off-university-top-tories-warn [accessed May 31, 2022].

² The one exception is Bavaria, where the incumbent government abandoned tuition to prevent loss in the upcoming election after an opposition party had successfully launched a publicity-boosting referendum against tuition. On the political controversies and the importance of public opinion for the abolishment of tuition in Germany, see, e.g., Times Higher Education, May 23, 2013, www.timeshighereducation.com/news/unbridled-success-germanys-fee-foes-claim-victory/2003928.article [accessed May 31, 2022] or Economy Watch, October 8, 2014, https://internationalviewpoint.org/spip.php?article4265 [accessed May 31, 2022].

In this paper, we conduct a series of randomized experiments to study how the choice of a regular vs. income-contingent design of the payment scheme affects the electorate's support to finance higher education through tuition. In the control group, we elicit public preferences for regular upfront tuition. In the baseline treatment group, we experimentally change the tuition payment scheme to a deferred income-contingent design.

We implement our analysis in a total of eight annual surveys with over 25,000 observations (plus a supplemental survey with N>2,000). Each annual survey is representative of the German voting-age population in terms of age, gender, parental status, education, and region. Fierce public debate surrounded the introduction and quick abandonment of tuition in Germany between 2006 and 2014, but income-contingent payment schemes barely entered the debate. Apart from studying the causal effect of payment schemes on public preferences for tuition, we implement additional experiments to explore issues of loan aversion, perceived political consequentiality, and various underlying mechanisms. Our repeated survey implementations also allow us to address replicability of the baseline experimental results, robustness in a within-subject design, and extensive treatment-effect heterogeneities. In addition, we field a separate survey of over 1,000 adolescents to study preferences of potential future university students.

The results of our baseline experiment indicate that designing tuition as income-contingent payments substantially increases support for tuition in the electorate. In the control group, 45.1 percent opposes regular upfront tuition and 44.0 percent favor it (the remainder is indifferent), indicating that the German public is strongly divided over whether or not to charge upfront tuition. The treatment that replaces the regular upfront design by a deferred income-contingent design raises support by 18.3 percentage points and reduces opposition by 18.8 percentage points. In the treatment group, a strong majority of 62.4 percent of the electorate favors income-contingent tuition and only 26.3 percent oppose it. The baseline finding consistently replicates in six separate experimental surveys.

While the experimental design allows for clean causal identification, in three separate surveys we implement a within-subject design that allows us to observe preferences for the two payment schemes within individuals. The descriptive within-subject results are very consistent with the experimental results and suggest that a quarter of the population opposes regular tuition while at the same time supporting income-contingent tuition.

In a separate experiment, we show that the treatment effect is robust to framing the incomecontingent payment scheme in the form of student loans. Debt aversion plays a role in many economic contexts including higher-education decisions (e.g., Loewenstein and Thaler, 1989; Oosterbeek and van den Broek, 2009; Field, 2009; Abraham et al., 2020). While our baseline treatment frames the financing scheme as deferred tuition payments, the literature often discusses upfront tuition payments combined with so-called income-contingent loans (ICLs) that involve the initial accumulation of debt. In an alternative experimental treatment, we make it highly salient that the scheme involves the initial take-up of loans: We frame the income-contingent form of tuition payoff-equivalently as tuition that is being paid in the form of income-contingent student loans. The estimated treatment effect is virtually identical to the baseline treatment. This result indicates that the preferences for income-contingent tuition are not affected by loan aversion and that respondents understand the underlying payment schemes and the intertemporal deferral of payment obligations.

In another separate experiment, we show that the treatment effect carries through to a setting where we experimentally increase respondents' perceptions about the political consequences of their survey answers. Public-opinion surveys have immense policy relevance (e.g., Blinder and Krueger, 2004; Hager and Hilbig, 2020). Still, a general potential concern with the use of stated preferences is that survey responses may be susceptible to reporting bias as they do not have immediate consequences (e.g., Carson, 2012; Kling et al., 2012). To validate our outcome measure, we experimentally inform respondents that average answers will be passed on to the responsible politicians in their state parliaments. We show that this treatment successfully augments respondents' perceptions about the political consequentiality of their survey answers but does not significantly change their stated preferences for upfront tuition or deferred incomecontingent tuition, preserving the large effect of the income-contingency treatment.

Subgroup analyses indicate that the treatment effect is particularly strong for groups who are likely to be represented in the political process. Politically active respondents (i.e., frequent voters and those who consider education topics important for their vote choice) and special interest groups (i.e., those with a university entrance qualification and parents) respond more strongly to the treatment, which provides additional relevance from a political-economy perspective. By contrast, treatment effects are quite homogeneous across various sociodemographic subgroups of the population.

The literature discusses various mechanisms why voters may prefer income-contingent payment schemes. If poor families are credit constrained, upfront tuition will deter their children from going to university (e.g., Carneiro and Heckman, 2002; Lochner and Monge-Naranjo, 2011). Deferred payment schemes that depend on students' future income streams alleviate such equity concerns. If future earnings streams are uncertain, income-contingent payment schemes also act as an insurance mechanism that encourages risky human capital investment (e.g., Findeisen and Sachs, 2016; Stantcheva, 2017). In addition, the uncertainty of

earnings expectations may be larger in some occupations than in others, restricting students' choice of subjects. A potential concern about income-contingent financing schemes is that they may require more sophisticated administration (e.g., Chapman, 2016). Income-contingent schemes may also have fiscal disadvantages compared to upfront schemes if some expenses cannot be recouped. In another experimental setup, we devise a battery of statements that allow us to experimentally test these various mechanisms.

Results of our mechanism analysis suggest that equity concerns and eased student situations are important reasons why people prefer income-contingent over upfront tuition. By far the biggest treatment effect on a mediator is that income contingency raises the share of respondents who perceive that people from poor families can afford to go to university despite tuition by 32.2 percentage points compared to upfront tuition. In addition, respondents are less likely to perceive income-contingent tuition as (i) unfair, (ii) deterring people from studying, (iii) restricting students' choice of subject and occupation, and (iv) increasing risk of indebtedness. By contrast, administrative and fiscal reasons do not seem to play a role. In a mediation analysis, about half of the overall income-contingency treatment effect on preferences for tuition can be attributed to the considered mediators, with altered fairness concerns accounting for more than one third.

When separating the two elements of payment deferral and income thresholds that underlie deferred income-contingent tuition, both show separate effects, but the former can virtually account for the entire overall effect. In a separate experiment, we add two experimental groups that are asked about tuition with deferred payment alone (but no income threshold) and about tuition above a threshold for (parental) income (but no payment deferral). While both of these elements significantly increase support for tuition compared to regular upfront tuition, payment deferral is the most important element driving public support. Interestingly, the sum of the two separate effects exceeds the combined effect, implying that the effects of the two elements are non-additive. Further experimental investigations show that other features of the income-contingency treatment such as reference to its implementation in other countries or explicit instruction to think of the same level of costs for the two types of tuition do not alter the income-contingency effect.

Finally, we show that income contingency also has a strong treatment effect on adolescents. As stakeholders in the education system, future voters, and potential future university students whose higher-education decisions are directly affected by financing schemes, adolescents are a specific population group with strong self-interests that plays a particular role in shaping higher education policy. To test whether the design of the payment scheme affects their preferences for tuition, we conduct a special survey of adolescents aged 14-17 years. While adolescents have much higher baseline opposition against regular upfront tuition (62.9 percent), the incomecontingency treatment reduces this opposition by 16.6 percentage points, doing away with the majority opposition. The treatment effect among adolescents is thus very similar to the votingage population. It is particularly large for those adolescents who expect to go to university, who are much more likely to oppose upfront tuition but not income-contingent tuition.

Our paper contributes to the literature on financing schemes for higher education. Incomecontingent payment schemes can increase both efficiency and equity by alleviating credit constraints and insuring against future income risk (e.g., Barr, 2004; Chapman, 2006; Jacobs and van der Ploeg, 2006; Chapman, Higgins, and Stiglitz, 2014; Dynarski, 2014; Findeisen and Sachs, 2016; Stantcheva, 2017; Diris and Ooghe, 2018; Barr et al., 2019). To date, however, there is no empirical evidence on how the design of tuition payment schemes affects the political feasibility of charging tuition. We advance the literature on the human capital, equity, and fiscal implications of tuition schemes with a political-economy perspective, stressing the large effect that financing schemes have on the electorate's preferences.

Thereby, we also contribute to the literature on the political economy of education. Political economy models study which population subgroups prefer public vs. private financing schemes (e.g., Fernandez and Rogerson, 1995; Epple and Romano, 1996; De Fraja, 2001; Glomm, Ravikumar, and Schiopu, 2011). Incorporating the choice among different financing schemes in political economy models, Borck and Wimbersky (2014) study theoretically the political support for income-contingent compared to other payment schemes. While these models tend to stress the opposing preferences of different population subgroups, our empirical results suggest that income-contingent payment schemes are preferred quite universally in the population. Quite generally, there is almost no empirical - and especially experimental knowledge about the determinants of electoral majorities for tuition and the political economy of higher education finance. As an exception, our companion paper (Lergetporer and Woessmann, 2023) studies experimentally whether information on the university earnings premium affects public preferences for regular tuition (but does not study deferred incomecontingent tuition or tuition payment schemes more generally). With our focus on education finance policies (as well as the expansion from adults to adolescents), we also extend the experimental literature on the public's preferences in various policy areas (e.g., Cruces, Perez-Truglia, and Tetaz, 2013; Kuziemko et al., 2015; Bursztyn, 2016; Alesina, Stantcheva, and Teso, 2018; Lergetporer et al., 2018, 2020; Haaland and Roth, 2020, 2023; Settele, 2022; Roth, Settele, and Wohlfart, 2021; see Haaland, Roth, and Wohlfart, 2023 for a survey).

The paper is structured as follows. Section 2 describes the institutional background, experimental setup, and empirical strategy. Section 3 presents our baseline results and various extensions. Section 4 analyzes two dimensions of potential underlying mechanisms. Section 5 presents results for adolescents. Section 6 concludes.

2. Background, Study Setup, and Empirical Strategy

This section describes the institutional background (section 2.1), our opinion survey (section 2.2), the experimental design of the baseline experiment (section 2.3), the empirical model (section 2.4), and the balancing of treatment groups (section 2.5).

2.1 Institutional Background

Germany's higher education system is mostly made up of public universities that are funded by the states. After a federal constitutional ban on tuition had been lifted, seven of the sixteen states introduced tuition of mostly 500 Euro per semester in the mid-2000s (see Bietenbeck et al., 2023). All seven states were run by a conservative government. The introduction of tuition was met by nationwide student protests and controversial legal and political debates with high media coverage (e.g., Dwenger, Storck, and Wrohlich, 2012). The topic played a prominent role in the campaigns of the subsequent election in all states. In the face of major public opposition, all states abandoned tuition within a few years, often as the first reform of a new administration after the tuition-introducing government got voted out of office in the next election (e.g., Kauder and Potrafke, 2013).

During the existence of tuition, students had to pay the tuition upfront while attending university. The concept of deferred income-contingent payment schemes did not play a major role in the public debate. The same is true for the experiences of prime examples of higher education finance systems where tuition is deferred through income-contingent loans such as Australia, England, and New Zealand (for details see, e.g., Chapman, 2016; Barr et al., 2019; Murphy, Scott-Clayton, and Wyness, 2019).³

While the past political-economy conflict around university tuition makes Germany a particularly interesting setting to study public support for tuition, it is not an international outlier in terms of its tuition policies. Germany's current policy of not charging tuition is found in seven of 27 countries analyzed by the OECD (2022) (Denmark, Estonia, Germany, Finland,

³ A related concept – income share agreements where tuition is paid as a percentage of later income – has been adopted, e.g., by Purdue University and an online learning start-up that raised major funding (New York Times, January 8, 2019, www.nytimes.com/2019/01/08/business/dealbook/education-student-loans-lambda-schools.html [accessed September 21, 2021]).

Norway, Sweden, and Turkey). Median annual tuition charged by public institutions for bachelor's degree programs is around USD 2,600, with the United States and England charging the highest tuition worldwide (12,255 and 9,212 dollars, respectively).⁴

2.2 The Opinion Survey

Our baseline analyses are implemented in seven waves of the ifo Education Survey (2014-2020), an annual opinion survey on education policy that we conducted in Germany. Each wave was carried out between April and July of the respective year and covered a nationally representative sample of the German voting-age population (18 years and older).⁵ Each year, respondents were asked to answer about 30 questions on different education topics. Median response time ranged between 15 and 24 minutes. The ifo Education Survey captures opinions across the entire education system, with a particular focus on the school system. Most waves presented roughly 25-35 substantive questions concerning preferences for various education policy topics. Each iteration consistently includes a core set of annual repeated questions (e.g., how respondents grade their state's school system or whether they support increased school spending). Besides these general questions, each wave contains an in-depth exploration of a specific focus topic. Examples include education policies to foster the integration of refugees (wave 2016), digitalization (wave 2017), and Covid-19 (wave 2020) (see Freundl et al. (2023) for details). The bulk of substantive policy questions in the survey utilize a five-point Likert scale format, offering choices like "strongly agree", "somewhat agree," "neither agree nor disagree", "somewhat disagree", and "strongly disagree". In addition to these substantive policy-specific questions, each survey wave also elicits comprehensive background data about respondents (usually 20-25 questions). This encompasses basic sociodemographics such as gender, age, education, occupation, and income, as well as aspects of respondents' economic preferences (e.g., risk and patience) and political preferences and behavior (e.g., party preferences and voting behavior). The ifo Education Survey data are available for scientific use (see Freundl et al. (2023) for information about data access). The questions regarding incomecontingent tuition were designed and posed specifically for the purposes of this study.

⁴ As in Germany, university tuition is also an important topic in the public discourse of other countries, and public protests often occur in response to the introduction or increase of tuition, e.g., in the UK (BBC, November 21, 2012, https://www.bbc.com/news/av/education-20432989), the United States (Washington Post, November 20, 2014, https://www.washingtonpost.com/news/post-nation/wp/2014/11/20/university-of-california-students-protest-tuition-hikes-a-look-at-some-campus-protests-over-the-years/), or Canada (CBC News, October 20, 2023, https://www.cbc.ca/news/canada/montreal/tuition-increase-montreal-protest-1.7012814).

⁵ Wave 2020 covered the population aged 18 to 69 years.

The surveys were implemented by the polling firms Kantar Public (formerly TNS Infratest, waves 2014-2019) and Respondi (wave 2020) who have access to nationally representative panels of adults. In the first four waves, stratified sampling was administered in two steps to account for the fact that a sizeable share of citizens (decreasing from 22 percent in 2014 to 17 percent in 2017) does not use the internet. Persons who use the internet were selected from an online panel and polled through an online survey. Persons who do not use the internet were polled at their homes, where trained interviewers provided them with tablet computers to complete the survey autonomously. All respondents answered all questions on a computer. This mixed-mode approach ensures that our sample is representative of the electorate, an aspect crucial to derive general political-economy statements (e.g., in the framework of median voter models). From wave 2018 on, the survey mode switched to fully online after extensive analysis revealed that representativeness for the entire population could be achieved through reweighting the online sample accordingly (Grewenig et al., 2023).

We employ survey weights calibrated to match administrative statistics with respect to age, gender, parental status, school degree, federal state, and municipality size. Inclusion of these weights does not substantially change the results of this paper, speaking towards the validity of our sampling strategy and the high quality of our data (see section 3.1).

Each wave included an item asking respondents about their preferences for university tuition. To experimentally study the effects of tuition payment schemes on participants' responses, respondents were randomly split into control and treatment groups. To ensure representativeness in each experimental group and sufficient power to detect economically relevant treatment effects, we generally aimed for 1,000 observations per experimental group.

In additional analyses, we also use data from the ifo Education Survey 2023. Wave 2023, conducted online in May and June 2023 by the polling firm Talk Online, had the same basic features as the previous waves and was again drawn and weighted to be representative for the population aged 18 to 69 years.

Table 1 shows the data structure across waves and treatment groups. In total, there are 24,333 observations in the adult population studied in this paper over the eight waves. As item

non-response is very low at around 1 percent on average,⁶ the overall sample size is 24,279 responses to the tuition questions.⁷

In wave 2018, we additionally conducted a separate adolescent survey with 1,085 respondents which we describe in greater detail in section 5.1 below.

Furthermore, in an auxiliary analysis we use data from an additional supplemental survey conducted online in March 2023 by the polling firm CINT/Gapfish. Its sample comprises 2,196 respondents drawn to be representative of the German adult population in terms of gender, age, and educational attainment.

2.3 Design of the Baseline Experiment

This section describes our baseline experimental design which implements a treatment about income contingency of tuition payments. In specific survey waves, we implement a series of additional analyses – a within-subject design, an additional experimental treatment on loan framing, an experimental validation with augmented perceptions of political consequentiality, and several analyses of mechanisms – which we describe in the respective sections below.

To test whether the design of the tuition scheme affects preferences for tuition, we devised a survey experiment that randomly assigns survey participants to a control group and a treatment group. Since our representative samples comprise respondents from various educational backgrounds, including nearly 40 percent with at most a basic school degree, we deliberately designed our survey instruments as simple as possible to minimize the possibility of comprehension problems.

In the control group, we elicited preferences for regular upfront tuition with the following question: "Do you favor or oppose that students at German universities cover a part of the costs of their studies themselves by tuition?" Respondents could pick one of five answer categories: strongly favor, somewhat favor, neither favor nor oppose, somewhat oppose, and strongly oppose.⁸

⁶ Whenever a question was left blank during the survey, a pop-up window with an encouragement to answer the respective question appeared. Respondents did not have the option to go back in the survey. Treatment status does not predict item non-response in our experiments (see Appendix Table A2).

⁷ The 24,279 observations come from 21,712 individuals, as some respondents participated in more than one wave. In regressions that pool different waves, we cluster standard errors at the individual level. Excluding respondents who participated again does not alter our results (see section 3.1).

⁸ Appendix Table A1 shows the wording of the different survey items. A summary description of the entire survey is available at www.ifo.de/ifo-bildungsbarometer. To prompt respondents to give a considered answer and to minimize the error of central tendency, the category "neither favor nor oppose" was placed below the other four answer categories. In methodological experiments on questions unrelated to tuition, we found that this reduces the error of central tendency and leaves the relative frequency of policy support and opposition unaffected. Other

To keep the questions on tuition as simple and general as possible, they did not specify the level of tuition. In Lergetporer and Woessmann (2023), we experimentally explore the impact of specifying a tuition level of 500 Euro per semester, a higher level of 1,500 Euro per semester, or not specifying any amount on support for regular upfront tuition. Support for tuition of 500 Euro is statistically not different from support without a specified tuition level, but support is significantly lower for tuition of 1,500 Euro. Thus, respondents have the level of 500 Euro per semester (i.e., the level charged in Germany before the abolition of tuition by 2014) in mind when they indicate their preferences for regular upfront tuition.

In the treatment group, respondents were asked whether they favor or oppose downstream, income-contingent tuition. Specifically, the question reads: "In other countries, there is tuition that is due only after graduation, when the former students earn income. Tuition has to be paid only if their annual income exceeds a certain threshold. Do you favor or oppose that students at German universities cover a part of the costs of their studies themselves by this alternative form of tuition?"

The experimental setting allows us to investigate whether preferences for tuition depend on whether tuition is designed as regular upfront payments or as deferred income-contingent payments. In contrast to regular tuition, income-contingent tuition was never introduced in Germany, and there were no major public debates about it. To convey that income-contingent tuition is a realistic policy option, we referred respondents to the fact that this form of tuition is in place in other countries.⁹

Our focus in this paper is whether the electorate's preferences for university tuition are influenced by the choice between a regular upfront or a deferred income-contingent design. The practicalities of the specific payment scheme chosen for income-contingent tuition involve many details such as the income threshold for repayment, repayment rate and duration, interest rate subsidy, and write-off period. Simulations of the impacts of income-contingent student loans suggest that the fine-tuning of these parameters affects the overall cost of higher education and its distribution among graduates (e.g., Britton et al., 2019). While recognizing the fiscal significance of these details, we deliberately refrain from experimentally manipulating these parameters. We posit that while voters can form preferences for the general concepts of regular and income-contingent tuition, many of them are unlikely to be able or willing to engage with

survey design details, such as the ordering and number of answer categories or their horizontal versus vertical presentation, also did not affect answering behavior (Woessmann et al., 2014).

⁹ Since we did not refer to other countries in the control group, a potential concern is that the effect of the income-contingency treatment might be driven by the reference to other countries. In section 4.3, we show experimentally that this is not the case.

these specific details. Given that approximately 80 percent of respondents have no university degree, these details are likely beyond the average respondent's immediate concern, making it improbable that they form preferences based on such specifics. This assertion is supported by our mechanism analysis (section 4.1) which shows that concerns about government or graduate costs associated with different tuition forms do not shape respondents' preferences. This finding suggests that the nuanced, cost-related parameters of income-contingent tuition are not driving our experimental results.

We first implemented the baseline experiment in wave 2015. Acknowledging the importance of replication to foster credibility of economic research (e.g., Maniadis et al., 2014), we implemented the same design again in waves 2016 and 2018-2020.¹⁰

2.4 Empirical Model

Our analysis of treatment effects is based on the following regression model:

$$y_i = \alpha + \beta T_i + \lambda' Z_i + \varepsilon_i \tag{1}$$

where y_i is individual *i*'s preference for tuition, T_i is an indicator of whether individual *i* was in the treatment group, Z_i is a vector of control variables, and ε_i is an error term that is uncorrelated with all right-hand-side variables. The parameter of interest is β , which represents the average treatment effect on preferences for tuition. Because of random assignment to the experimental groups, additional control variables are not required to identify the causal treatment effect. Nevertheless, their inclusion may increase the precision of estimates.¹¹

To analyze heterogeneity of the treatment effect across different subgroups of the population, we extend our basic regression model to:

$$y_i = \alpha + \beta_1 T_i + \beta_2 S_i + \beta_3 T_i S_i + \lambda' Z_i + \varepsilon_i$$
⁽²⁾

where S_i equals one if respondent *i* is member of the respective subgroup and zero otherwise. The treatment effect for non-members of the subgroup is given by β_1 , and β_3 measures the additional effect on the subgroup.

¹⁰ We registered our study (AEARCTR-0003873) *after* the first waves of data collection, in early 2019. This was due to the fact that the practice of preregistering economics experiments was not well-established when we started our project in 2014. Importantly, replicating experimental findings several times as we do mitigates any concerns regarding the results being false positives (Coffman and Niederle, 2015).

¹¹ See table notes for lists of included covariates. The share of missing covariate data is very low in our samples (below 1 percent, on average). Throughout the paper, we impute missing covariates by a constant and include dummies indicating imputed values for each covariate. Our qualitative results hold when observations with missing covariates are dropped (see section 3.1).

2.5 Tests of Randomization

To test whether randomization in our experiments successfully balanced respondents' characteristics across treatment and control groups, we investigate whether our rich set of covariates differs by treatment status. Columns 1-7 of Appendix Table A2 present means of observable characteristics for the control groups in the different waves. Columns 8-15 report coefficients and significance levels of regressions of the form

$$C_i = \alpha + \gamma T_i + \varepsilon_i \tag{3}$$

for each covariate C_i and each experiment separately.

It is reassuring that only 11 out of 264 regressions (4.2 percent) yield a coefficient γ that is significant at the 5 percent level, as would be expected by pure chance. Thus, the balancing tests suggest that random assignment worked as intended.

3. Results

This section presents our experimental results on how a deferred income-contingent design of the tuition payment scheme affects public support for tuition. We show our baseline results and their replicability (section 3.1), report analyses within individual subjects (section 3.2), study how support for income-contingent tuition depends on the absence or use of a loan framing (section 3.3), validate that stated preferences are not biased by lack of perceived political consequentiality (section 3.4), and provide evidence on heterogeneous treatment effects (section 3.5). The subsequent sections 4 and 5 turn to analyses of mechanisms and preferences of adolescents, respectively.

3.1 Baseline Results and Replication

Our baseline experiment studies whether the public's preferences for whether or not students should cover part of the university costs themselves depends on the regular vs. deferred income-contingent design of tuition payments. The control group is asked about their preferences for the regular form of tuition that is paid upfront when students attend university. The treatment group is asked about their preferences for income-contingent tuition where payments are deferred until the former students earn sufficient income.

Table 2 shows the estimated effects of designing tuition as deferred income-contingent payments on preferences for tuition based on equation (1). To depict political majorities for or against tuition, the first two columns focus on whether respondents are in favor of tuition (combining the answer categories "strongly favor" and "somewhat favor"), whereas the latter

two columns focus on opposition against tuition (combining "somewhat oppose" and "strongly oppose"). The residual answer category on the five-point scale is "neither favor nor oppose."¹² Estimates in columns 1 and 3 do not include covariates, whereas columns 2 and 4 include standard covariates as listed in the table notes.

Control-group results indicate that the German population is divided about charging upfront tuition. In the first experimental wave (2015), 43.8 percent of respondents in the uninformed control group favor tuition and 46.4 percent oppose tuition (see control mean in Table 2); the remaining 9.8 percent are undecided.

The experimental treatment shows that designing tuition as deferred income-contingent payments substantially increases public support, creating a strong majority of the electorate in favor of tuition. The income-contingency treatment raises support for tuition by 15.3 percentage points in wave 2015 (column 1 of Table 2). Concurrently, opposition against tuition declines by 15.8 percentage points (column 3).

The results replicate very closely in four other representative samples in waves 2016, 2018, 2019, and 2020. By stacking the control and treatment groups of the five waves, Table 2 allows for a direct comparison of results across waves. Support for regular tuition in the control group does not differ significantly across the waves. The income-contingency treatment effect is almost identical in the first three waves and (marginally) significantly larger in the final two waves.¹³ Baseline level and treatment effects on opposition against tuition are also similar across waves, with the treatment effect in the final three waves slightly (but not significantly) larger. The replicability indicates that the highly significant effect of the income-contingency treatment on preferences for tuition does not reflect a false-positive result but rather the true causal effect of the treatment on public preferences (see Maniadis et al., 2014).

Figure 1 shows the pooled results of the experiment across the five waves. On average, the income-contingent design increases support for tuition by 18.3 percentage points and reduces opposition by 18.8 percentage points. The treatment turns a minority supporting tuition in the

¹² All regression results presented in this paper are estimated using weighted OLS regressions. Non-weighted regressions (Appendix Table A3) and nonlinear probit models (Appendix Table A4) yield qualitatively identical results. Results are also robust to excluding respondents who participated in more than one survey wave (Appendix Table A5) and excluding observations with missing covariates (Appendix Table A6).

¹³ When pooling waves 2015-18 and waves 2019-20, the difference in the treatment effect on support for tuition between these two periods is statistically highly significant. The structural break between 2018 and 2019 does not align with any methodological changes made to the ifo Education Survey over time: the transition from a mixed-mode to an online survey occurred in 2018 and a change in survey providers in 2020 (see section 2.2). Neither do any other obvious modifications in survey content coincide with the break. Although we cannot identify a specific cause for the one-time shift, it is important to note that the treatment effect is both substantial and statistically highly significant in every single survey wave, fluctuating between 15 and 22 percentage points.

regular design into a strong majority favoring tuition in the income-contingent design. On average, 62.4 percent of participants are in favor of having tuition that is due only after graduation and only if the former students earn income above a certain threshold. Only 26.3 percent of the population oppose this form of tuition.

Inspection of treatment effect estimates on each of the five underlying answer categories indicates that the income-contingency treatment shifts strongly held preferences for and against tuition as well as less strongly held preferences. Results using the full preference elicitation on a five-point scale in Appendix Table A7 indicate that public opinion is shifted throughout the spectrum.

3.2 Within-Subject Design

To compare preferences for the two tuition payment schemes within individuals, in two other waves (2014 and 2017) we elicited preferences for upfront tuition and for incomecontingent tuition in two separate consecutive questions. The question on income-contingent tuition is placed on a separate screen directly after eliciting preferences for regular tuition. Both questions are worded identically to the questions used in the experimental treatments of the other waves. While the experimental treatment in the five waves considered so far (2015, 2016, and 2018-2020) allows us to estimate treatment effects of the upfront vs. income-contingent tuition designs, the within-subject setup allows us to perform within-respondent comparisons of preferences for upfront and income-contingent tuition. As this consecutive and non-experimental elicitation could in principle bias responses on the second question, e.g., if there is preference for giving consistent answers (e.g., Falk and Zimmermann, 2013), the main analysis so far was restricted to the experimental elicitations in the other waves.

Table 3 shows the distribution of individuals' preference profiles for the two types of tuition. Overall, a majority of 64.2 percent of respondents favor income-contingent tuition in this setting, compared to 41.5 percent favoring regular tuition. The non-experimental elicitation of preferences for income-contingent tuition after eliciting preferences for regular tuition thus provides similarly large support for income-contingent tuition as in the experimental elicitation in the other five waves. In fact, support in the non-experimental setting is slightly higher, suggesting that prior consideration of regular tuition might induce a few participants to look more positively at income-contingent tuition (although wave effects may also play a role).

The main diagonal of Table 3 reveals that the majority of respondents (55.2 percent) has the same preferences for both types of tuition. The areas off the main diagonal indicate that only 9.2 percent are more supportive of regular than income-contingent tuition, whereas 35.7 percent

are more supportive towards the latter. The biggest difference comes from 25.2 percent of respondents who oppose regular tuition but are in favor of income-contingent tuition. These respondents stand out due to their responsiveness to deferred income-contingent payments. To describe them, Appendix Table A8 regresses a dummy variable capturing their preference profile on respondent characteristics. Holding a university-entrance qualification and living in a state with tuition in place in the mid-2000s are the only variables significantly (and positively) correlated with this preference profile.

Waves 2014 and 2017 also include an experiment that provides a treatment group with information on the university earnings premium (studied in our companion paper, Lergetporer and Woessmann, 2023). The earnings information is provided on the survey screen that elicits preferences for regular tuition, and preferences for income-contingent tuition are elicited on the next survey screen (the within-subject analysis in Table 3 is based on the control groups that did not receive any information). The consecutive elicitation of preferences for regular and income-contingent tuition allows us to estimate the causal effect of the earnings information treatment on preferences for income-contingent tuition.¹⁴ Results shown in Appendix Table A9 do not suggest that the earnings information treatment has a noteworthy effect on preferences for income-contingent tuition. Preferences for income-contingent tuition thus seem less malleable to providing information on relative earnings than preferences for regular tuition.

3.3 Loan Framing

While the income-contingent version of tuition entails an intertemporal deferral of payments, the baseline formulation of the income-contingency treatment does not include an explicit reference to loans. Income-contingent tuition is characterized as a deferred payment, rather than as an upfront payment obligation combined with taking up an income-contingent loan (ICL) with an equivalent repayment scheme (as implemented in Australia, England, and New Zealand). While these two characterizations refer to equivalent payment schemes, public preferences may differ between the two characterizations if people are subject to loan aversion (e.g., Loewenstein and Thaler, 1989; Oosterbeek and van den Broek, 2009; Field, 2009;

¹⁴ The analysis requires the assumption that respondents memorize the information provided on the prior survey screen when answering the subsequent question on income-contingent tuition. This assumption seems reasonable because the effects of the earnings information treatment on knowledge of the earnings information and on preferences for regular tuition persist in a follow-up survey two weeks later (Lergetporer and Woessmann, 2023).

Abraham et al., 2020). It is often suggested that Germans may have a particularly strong aversion to taking up loans.¹⁵

To test whether preferences for income-contingent tuition depend on loan aversion, we devised another experiment in wave 2018 that assigned participants to one of three experimental groups. The design of the control group and the first treatment group are the same as in the baseline experiment. Thus, respondents in the first treatment group were asked about income-contingent tuition using a wording that did not refer to loans but indicated a deferred payment of tuition. In the second treatment group, respondents were asked effectively the same question as in the first treatment group, but in a version that framed this form of tuition payoff-equivalently as tuition that is being paid in the form of income-contingent student loans: "In other countries, there is tuition that is being paid in the former students earn income. The student loans only have to be paid if their annual income exceeds a certain threshold." By explicitly referring to "loans" three times, this alternative treatment aims to test whether loan aversion affects results.

Results indicate that the strong appeal of income-contingent tuition is unaffected by making its debt aspect salient through the loan framing that explicitly refers to loan repayment requirements. Table 4 shows that the loan-framing treatment has virtually the identical effect on public support for tuition as the baseline income-contingency treatment. Even across the five underlying answer categories, there are no statistically significant differences between the two treatment groups (Appendix Table A10). Thus, the majority appeal of income-contingent tuition is unaffected by making its debt aspect salient through an explicit loan framing, indicating that respondents understand the underlying payment schemes and the intertemporal deferral of payment obligations. This finding is particularly interesting in light of the relevance that debt aversion seems to have for students' preferences for income-contingent loan repayment schemes (see Abraham et al., 2020, for survey evidence from undergraduates at the University of Maryland).

3.4 Perceived Political Consequentiality

Our baseline results are based on stated preferences for tuition. A general concern with using stated preferences as outcome measures is that they may be susceptible to reporting bias because

¹⁵ For examples, see BBC, February 13, 2015, www.bbc.com/news/business-31369185 [accessed September 21, 2021] or Handelsblatt, January 26, 2018, www.handelsblatt.com/today/opinion/germans-and-money-germanys-investment-culture-has-the-makings-of-a-social-and-financial-crisis/23580894.html [accessed September 21, 2021].

they carry no immediate political consequences (e.g., Carson, 2012; Kling et al., 2012). Reassuringly, some papers in the literature have validated self-reported survey responses with actual political behavior such as donating money or signing petitions (e.g., Alesina, Miano, and Stantcheva, 2023; Grigorieff, Roth, and Ubfal, 2020; Haaland, Roth, and Wohlfart, 2023; Settele, 2022).

In the spirit of this strand of research, we conducted another experiment in wave 2020 to examine whether our experimental results on preferences for tuition hold when we experimentally increase respondents' perception that their answers carry political consequences. We randomly assigned respondents to one of four experimental groups. The control group and the first treatment group answered the questions about preferences for regular and income-contingent tuition, respectively, in the same way as in the baseline experiment. In the other two experimental groups, respondents were first informed on a separate screen that their aggregate answers would be passed on to their state-parliament politicians. After receiving this information, they stated their preferences for regular and income-contingent tuition, respectively. This two-by-two experimental design does not only provide causal evidence on how increasing perceived political consequentiality affects the level of stated preferences for regular and income-contingent tuition, respectively. It also reveals how perceived political consequentiality affects the income-consistency treatment effect, i.e., the *difference* in public preferences for the two types of tuition. After the survey was completed, we informed each of the 157 education-policy spokespersons in all state parliaments by email about the share of survey respondents favoring regular and income-contingent tuition.

Columns 1-4 of Table 5 show that increasing perceived political consequences to the survey responses does not affect stated preferences for regular and income-contingent tuition or the difference between the two. Among respondents asked about regular upfront tuition, the perceived-consequentiality treatment slightly increases preferences for regular tuition by an insignificant three percentage points. The income-contingency treatment effects without and with increased perceived consequentiality are very similar at 22.1 and 20.5 percentage points, respectively, and the difference is not statistically significant (p>0.48, post-estimation Wald test). These results alleviate concerns that the lack of perceived political consequences of responses to the tuition questions in the baseline experiment could lead to reporting biases and thereby produce a distorted picture of true underlying preferences for tuition.¹⁶

¹⁶ Relatedly, recent research shows that experimenter-demand effects are largely absent in survey experiments (Mummolo and Peterson, 2019; Grewenig et al., 2022) and in experiments on economic preferences (de Quidt, Haushofer, and Roth, 2018).

In our supplemental survey 2023, we verify that the perceived-consequentiality treatment indeed shifts respondents' perceptions regarding the political consequences of their survey answers. Respondents were randomly divided into either a control group or the perceivedconsequentiality treatment group, which was informed that their average answers will be passed on to their state politicians. After eliciting their preferences for income-contingent tuition, we asked them to what extent they agree to the statement "My answer to the previous question about tuition can impact education policy in my state", recording answers on a five-point scale from fully agree to fully disagree. Replicating our result from wave 2020 of the ifo Education Survey, the perceived-consequentiality treatment does not affect preferences for incomecontingent tuition (see columns 5-6 of Table 5). Notably, however, it does significantly increase agreement to the statement that the survey response can impact state education policy by 9 percentage points (columns 7-8). This finding corroborates the notion that the absence of impact of the perceived-consequentiality treatment on preferences for tuition emerges because perceived political consequences of survey answers do not change stated preferences, not because the perceived-consequentiality treatment does not have an effect on perceived consequentiality (see also Lergetporer and Woessmann, 2023).

3.5 Subgroup Analysis: Sociodemographics, Political Activity, and Interest Groups

Next, we test whether the substantial average effect of the income-contingency treatment on preferences for tuition masks important heterogeneities across subgroups of respondents. We study the extent to which treatment effects vary by sociodemographic characteristics and then focus in particular on politically active respondents and members of interest groups, as their preferences may be especially relevant for the political feasibility of implementing tuition reforms.¹⁷

To set the stage for the heterogeneity analysis, we first provide descriptive evidence on how preferences for tuition vary by background characteristics. Appendix Table A11 presents OLS regressions of support for regular and income-contingent tuition on respondents' characteristics using the stacked data of all waves. Each cell in columns 1 and 3 presents the coefficient of a bivariate regression of preferences for the respective form of tuition on the respective characteristic, and columns 2 and 4 each present a multivariate regression that includes all characteristics simultaneously. Starting with preferences for regular tuition as dependent

¹⁷ Note that the subgroup analysis is exploratory, aimed at identifying variation in effects across different population subgroups rather than testing predetermined hypotheses. Therefore, we refrain from strong substantive interpretations of the observed patterns.

variable in columns 1 and 2, partisans of the conservative party (CDU) which had introduced tuition in Germany are more likely to support tuition (see columns 1 and 2). Furthermore, individuals with a university entrance qualification (i.e., potential students), current university students, and to a lesser extent also university graduates and parents are less supportive, suggesting the relevance of pecuniary self-interest in explaining preferences for tuition. Furthermore, males, those working in the education sector, and full-time employees are more likely to support regular tuition.

The correlational patterns are rather similar when considering preferences for incomecontingent tuition in columns 3 and 4, although some interesting differences stand out. Again, CDU partisans are more likely to prefer this form of tuition, and current university students and university graduates are less likely to support income-contingent tuition. The association between holding a university entrance qualification and support for tuition is much smaller for income-contingent than regular tuition. In contrast to regular tuition, frequent voters are more likely to support income-contingent tuition. In addition, support for income-contingent tuition is positively associated with age and full-time employment.

To investigate treatment-effect heterogeneities across different subgroups, Table 6 reports regression results based on equation (2). Column 1 depicts the main treatment effects for the respective omitted subgroup (i.e., β_1 of equation (2)), and column 2 depicts the coefficients on the treatment-subgroup interaction (i.e., β_3 of equation (2)). Depending on the respective characteristic, the sample is split into two or three subgroups.

Treatment effects are positive and significant in each subgroup considered in column 1 of Table 6, and only six of the 19 interaction terms are statistically significant at the 5 percent level (see column 2). This pattern suggests that the income-contingency effect reported in the previous sections is rather general and not due to extreme reactions of subgroups with certain sociodemographic characteristics.

At the same time, the table reveals some interesting heterogeneities. From a politicaleconomy perspective, it is noteworthy that treatment effects are significantly larger for respondents who are politically active, i.e., frequent voters and respondents who consider education topics important for their voting decision. As these subgroups are more likely to be represented in the political process, this result indicates the political feasibility of implementing income-contingent tuition.

Turning to special interest groups, we find that treatment effects are significantly larger for persons who hold a university entrance qualification and for parents with children aged above

18 years.¹⁸ Special interest groups like these may invest resources to influence voting outcomes in their favor (e.g., Grossman and Helpman, 2001), which implies that they may play a particular role in determining higher education finance (Glomm, Ravikumar, and Schiopu, 2011). Furthermore, younger respondents (below age 45) respond less strongly to the treatment, whereas women respond slightly more strongly.

A final heterogeneity result shows that our main results are not driven by respondents' attention levels. We measured respondents' attentiveness in the 2019 wave of the ifo Education Survey.¹⁹ Results show that the income-contingency effect is substantial and statistically significant for both attentive and inattentive respondents (Appendix Table A12). As one would expect, the effect is significantly stronger among attentive respondents (+23.9 percentage points for support and -23.8 for opposition) compared to inattentive respondents (+14.2 for support and -11.4 for opposition).

The heterogeneity analysis supports the finding that replacing regular upfront payments with deferred income-contingent payments fosters the political feasibility of charging tuition. Beyond the higher average support for income-contingent tuition in the population, the treatment particularly affects the preferences of politically active subgroups that may play a crucial role in the policy-making process. Still, the heterogeneity analyses should be interpreted with caution, as statistical power to identify subgroup heterogeneity is somewhat limited despite the large sample size.

4. Analysis of Mechanisms

This section presents additional experimental survey designs to study two dimensions of mechanisms that may lie behind the pronounced effect of the income-contingency treatment on support for tuition. First, we investigate possible mediators of the treatment effect by studying how income contingency affects respondents' perceptions of how tuition impacts fairness, students' human-capital investment decisions, and administrative and fiscal implications (section 4.1). Second, we study the relative importance of payment deferral and income thresholds as the two separate elements of income-contingent tuition (section 4.2). In addition, we explore whether heightened support for income-contingent tuition might arise from the

¹⁸ There are relatively few current university students (who would have to pay tuition if it was introduced) in the sample, leading to very imprecise subgroup estimates.

¹⁹ We employed the following attention-check question, similar to Settele (2022): "It sometimes happens that respondents do not read individual questions carefully. To ensure that you read the questions carefully, we ask you to ignore the following question and enter the number twenty-two in the text field." Respondents who entered "22" are categorized as attentive (76.9 percent), the others as inattentive.

reference to its implementation in other countries or from differences in the costs that respondents associate with the two types of tuition (section 4.3).

4.1 Changed Perceptions about Tuition as Mediators

Experimental Design. To explore possible reasons for the difference in support for the regular upfront tuition payment scheme and the deferred income-contingent scheme, we devised a battery of statements in wave 2019 to measure variables that might act as mediators. We again randomly divided respondents between a control group and the income-contingency treatment group. On a first survey screen, we elicited their preferences for tuition using the same design as in the baseline experiment. On a subsequent survey screen, respondents were asked to provide their assessment of a battery of nine different statements about tuition. Respondents in the control group assessed statements about regular upfront tuition; respondents in the treatment group assessed the same statements about deferred income-contingent tuition. Guided by the theoretical literature on differences between regular and income-contingent repayment schemes (e.g., Chapman, Higgins, and Stiglitz, 2014; Barr et al., 2019), we study three sets of possible mechanisms for why respondents may favor income-contingent tuition.

The first set of factors relates to fairness concerns. Upfront tuition may discourage children from poor families from attending university when credit constraints are binding (e.g., Carneiro and Heckman, 2002; Lochner and Monge-Naranjo, 2011). Because deferred income-contingent tuition depends on students' future income streams, it alleviates such concerns. To capture such types of fairness concerns, we elicited respondents' agreement with the following three statements: "Also people from poorer families can afford to study despite tuition." "Tuition is unjust." "Through tuition, future high earners contribute to the costs of studying." For each of these (and the following) statements, respondents could pick one of five answer categories: fully true, rather true, neither nor, rather not true, and not true at all.

The second set of factors refers to concerns about tuition-induced distortions in (prospective) students' human-capital investment decisions. Income-contingent payment schemes encourage human-capital investment because they act as an insurance mechanism against uncertainty of future earnings (e.g., Findeisen and Sachs, 2016; Stantcheva, 2017). In contrast, regular tuition can increase students' risk of debt, restrict their choices of study subject and careers to higher-paying occupations to repay tuition debt, or even discourage people from studying altogether. The three statements that measure concerns about (prospective) students' educational choices are as follows: "Tuition deters people from studying." "Tuition restricts

students' choice of subject and profession." "Tuition increases the risk of students getting into debt."

The third set of factors refers to administrative and fiscal considerations. One potential criticism of income-contingent tuition is that it may be harder to administer (e.g., Chapman, 2016). In addition, there may be fiscal disadvantages compared to regular tuition if some expenses cannot be recouped (e.g., because graduates move abroad), which may also mitigate positive effects of tuition revenues on university quality. We measure these concerns by asking respondents to assess the following statements: "Tuition is difficult for the state to administer." "Through tuition, the state is relieved of financial burden." "Tuition income enables universities to improve their quality."

Treatment Effects on Potential Mediators. A necessary condition for variables to act as mediators is that they must be significantly affected by the treatment. As a first step, we therefore study the extent to which the income-contingency treatment affects respondents' assessments of the battery of statements about tuition.

Results in Table 7 show that the public perceives income-contingent tuition as fairer and less detrimental for (prospective) students than regular tuition. Most strikingly, while only 33.0 percent in the control group think that people from poorer families can afford to study despite regular upfront tuition, this share increases by 32.2 percentage points in the treatment group that considers deferred income-contingent tuition (column 1). That is, a key difference in the perception between the regular and the income-contingent design of tuition is that the share of the public who expect that tuition makes studying impossible for the poor roughly doubles to two thirds in the latter case. Asking directly about fairness, the share of those who consider tuition (column 2). And while agreement that tuition makes future high earners contribute to the costs of their studies is already high for regular tuition at 70.4 percent, this increases even further by 6.2 percentage points when specifying income-contingent tuition (column 3).

Focusing on perceived tuition effects on (prospective) students, majorities of control-group respondents think that regular tuition deters people from studying (column 4), restricts students' choice of subject and profession (column 5), and increases the risk of student indebtedness (column 6). These shares are substantially reduced by 14.1 to 16.2 percentage points in the treatment group that considers income-contingent tuition.

By contrast, there are no marked differences in how the two types of tuition are perceived to affect the government. Perceptions about administrative difficulties (low at 26.5 percent, column 7) and relieve of government financial burden (high at 65.8 percent, column 8) are not

statistically or economically significantly affected by the income-contingency treatment. Perceptions that tuition enables universities to improve their quality are somewhat higher by 7.8 percentage points for income-contingent than for regular tuition, but both are on a high level (column 9).

To shed some light on how the design of the repayment scheme affects another determinant of the fiscal impact of charging tuition – preferred tuition levels – we asked two additional randomly selected subgroups of respondents about their preferred maximum tuition level (see Appendix Table A1 for question wording). Among those who do not oppose tuition in the first place, the median response in the control group is 300 Euro of regular upfront tuition per semester. In the treatment group, the median is significantly higher at 500 Euro of deferred income-contingent tuition per semester (p<0.01). This latter level is exactly the rate that had been charged in Germany before tuition was abolished. This result indicates that incomecontingent payment schemes also raise the fiscal scope for tuition, as the public accepts higher levels of tuition in the deferred income-contingent design.

Overall, this analysis suggests that reduced concerns of adverse effects on fairness and (prospective) students may be important mechanisms by which the income-contingent design of the payment scheme increases public support for tuition.²⁰ It is remarkable that respondents' perceived advantages of income-contingent tuition, such as lower risk of deterring (poor) students or of student indebtedness, are very much in line with early arguments in favor of income-contingent payment schemes (e.g., Friedman, 1955). At the same time, the public does not seem to share the concern that deferred repayment options are harder to administer (e.g., Barr et al., 2019).

Mediation Analysis. We can use these results on treatment effects on mediators to conduct a mediation analysis to quantify the share of the overall income-contingency treatment effect on preferences for tuition that can be attributed to these mediators. We follow the approach developed by Heckman, Pinto, and Savelyev (2013) and Heckman and Pinto (2015) which decomposes the overall treatment effect into a share explained by k observed mediator variables and a remaining share attributed to unobserved mediators.

Assuming that the outcome can be expressed as a linear combination of mediator variables and sociodemographic control variables, the approach extends the baseline estimation model from equation (1) to:

²⁰ Appendix Table A13 shows that respondents' perceptions about tuition are meaningfully associated with their preferences for regular and income-contingent tuition.

$$y_i = \alpha + \beta^{residual} T_i + \sum_k \theta^k M_i^k + \lambda' Z_i + \varepsilon_i$$
(4)

where $\beta^{residual}$ represents the treatment effect not explained by changes in the observed mediators. Thus, the share of the treatment effect that is explained by the combined changes in the observed mediators is given by $1 - \beta^{residual}/\beta$ (with β from equation (1)).²¹ To calculate the share of the overall treatment effect attributable to the k^{th} mediator, the approach multiplies the treatment effect on the respective mediator (reported in Table 7) by the mediator's association with the outcome θ^k and divides by the overall treatment effect β . The relevance of a mediator thus depends both on how strongly it is affected by the treatment and on how strongly it is related to the outcome.

Table 8 depicts the results of the mediation analysis, focusing on the seven mediators that are significantly affected by the income-contingency treatment. By far the most important set of mediators are fairness concerns. Panel A considers each mediator separately. Agreement with the statements (i) that people from poor families can afford to study despite tuition, (ii) that tuition is unjust, and (iii) that tuition makes future high earners contribute account for 36.0 percent, 21.8 percent, and 8.0 percent of the overall income-contingency treatment effect, respectively (columns 1-3). Perceived tuition effects on (prospective) students – whether it deters people from studying, restricts students' choices, or increases their debt risk – are also relatively important, accounting for 12.7 to 13.7 percent each (columns 4-6). In contrast, perceptions that tuition improves universities' quality are relatively less important, accounting for only 8.0 percent.

Considering all mediators jointly in Panel B shows that changes in fairness concerns account for 34.9 percent of the overall treatment effect, changes in concerns about (prospective) students for 11.7 percent, and changes in perceptions about the effects of tuition on university quality for 4.3 percent.²² All mediators together thus account for 50.9 percent of the income-contingency effect on preferences for tuition, which is a substantial share for a mediation analysis given limits to observing mediators.

²¹ The implicit assumption in this setup is that all potential unobserved mediators subsumed in the error term are orthogonal to the included mediator and control variables.

²² Interestingly, indicators of respondents' general altruistic inclination, such as employment in the education sector (Dohmen and Falk, 2010) or direct survey-based measures of altruism (Falk et al., 2023), do not correlate with support for income-contingent tuition in our data. Therefore, the prominent role of fairness concerns in the mediation analysis likely reflects specific concerns about the fairness of tuition rather than general other-regarding preferences.

4.2 Payment Deferral and Income Thresholds as Separate Elements

Experimental Design. The established importance of fairness considerations raises the question which specific features of income-contingent tuition are pivotal to garner their support. Importantly, income-contingent tuition, as described in the baseline experiment, differs from regular tuition in two notable ways. First, there is the element of payment deferral, meaning that tuition is due only after graduation, when the former students no longer go to university and can focus on working on the labor market. Second, there is the element of an income threshold, meaning that payments are only due if the graduates' incomes surpass a certain cutoff. For tuition to be contingent on students' own future income, both elements have to come together to define income contingency. But it is worth exploring whether one of these elements dominates the other in accounting for the greater popularity of income-contingent tuition.

To test which of the two elements elevate support for income-contingent tuition, we implemented another experiment in wave 2023 of the ifo Education Survey. We randomly assigned respondents to one of four experimental groups. In the first two groups, we replicated the questions regarding regular upfront and income-contingent tuition from our baseline experiment. The third experimental group was asked about tuition with deferred payment, but without an income threshold. The fourth group was asked about tuition with an income threshold, but without deferred payment. Given that upfront tuition without deferred payment cannot be based on students' future income, we specified in this group that tuition is due only if students' parents earn above a certain income level (see Appendix Table A1 for the question wording).²³

Results. Table 9 presents the impacts of the different versions of tuition on respondents' support. Replicating our prior results, the income-contingency treatment strongly increases support for tuition by 13.8 percentage points (column 1) and curtails opposition by 17.5 percentage points (column 3).

Importantly, tuition with deferred payment, but without an income threshold, produces effects nearly as large as the income-contingency treatment. The payment-deferral-alone treatment increases support by 11.2 percentage points and decreases opposition by 16.9 percentage points. The differences to the treatment effects of the baseline experiment are not

²³ Appendix Table A14 shows that respondents' characteristics are balanced across experimental groups in the 2023 survey.

statistically significant. Thus, payment deferral seems to be the driving force behind the increased support for deferred income-contingent tuition.²⁴

By contrast, effects of the income-threshold-alone treatment (without payment deferral) are significantly smaller than the effects of the baseline treatment. While implementing an income threshold without deferred payment also significantly increases support for tuition relative to regular upfront tuition, its effects amount to a smaller increase in support of 7.3 percentage points and a 11.8 percentage-point decrease in opposition.²⁵ That is, implementing a payment threshold for parental income also raises support for tuition, but not to the same extent as deferred tuition that is contingent on students' own future income.

Interestingly, the sum of the two individual effects of payment deferral and income threshold exceeds the effect of the combination in the income-contingency treatment. This result implies that the effects of the two elements are non-additive.

In sum, the experimental results in this section reveal that the deferral of repayment is the key factor driving public support for income-contingent tuition over regular tuition. Income thresholds also seem to contribute to a more positive view of charging tuition, but the fact that thresholds in upfront tuition have to refer to parental income limits a full assessment of the relative importance of the two elements. Still, the particular importance of deferral highlights the broader applicability of our findings beyond the German context, given that numerous countries including the U.S. offer tuition models with student loan systems that allow for deferred payments – a feature that seems vital for gaining public backing.

4.3 Reference to other Countries and Explicit Constancy of Costs

We also test whether two additional features – the reference to other countries and the absence of a statement of comparable costs – can account for our baseline treatment effect.

Reference to other Countries. In the baseline experiment, we mention that incomecontingent tuition exists in "other countries", which could affect public preferences as it hints to international precedents. To test the relevance of the reference to other countries, we added an additional group to the experiment in wave 2023 of the ifo Education Survey. This

²⁴ The large effects of mentioning deferred payments (without income thresholds) compared to the control statement imply that respondents in the control group have upfront tuition in mind when answering the question on regular tuition (which does not explicitly mention upfront payment).

²⁵ The fact that upfront tuition based on parental income is significantly more popular than upfront tuition without income contingency underlines that the common practice in many countries to couple tuition with needs-based support programs (like the Pell Grant in the U.S.) likely increases support for tuition.

experimental group was presented the standard question about income-contingent tuition, but without mentioning "other countries" (see Appendix Table A1 for the question wording).

Reassuringly, omitting the mention of "other countries" leaves the effects of the incomecontingency treatment virtually unchanged. As shown in the final row of Table 9, treatment effects are close to the baseline experiment, and the differences are not statistically significant, raising confidence that the treatment captures genuine effects of income contingency.

Explicit Constancy of Costs. The baseline experiment does not explicitly detail the associated costs of the two forms of tuition. This vagueness might lead respondents to base their preferences on different implicit cost assumptions. To test for the relevance of this concern, we additionally designed an integrated within-subject experiment in wave 2023 to hold constant the costs that respondents have in mind when stating their preferences for the two forms of tuition. In particular, respondents initially randomized into the first experimental group that elicited preferences for regular tuition were subsequently posed a question on incomecontingent tuition. This question was coupled with the instruction to "think of total tuition costs for the former students of the same amount as for regular tuition in the previous question."²⁶ This approach enables us to discern preferences for both types of tuition while explicitly holding constant the perceived costs.

Results show that differing cost perceptions between regular and income-contingent tuition do not drive the latter's higher support. When costs are held constant between the two tuition types, 62.5 percent of respondents support income-contingent tuition, but only 46.0 percent of the same respondents support regular tuition (see Appendix Table A15). In line with our previous within-subject results (section 3.2), the within-subject difference in support between regular and income-contingent tuition is even slightly larger than in the between-subject experiment, implying that differential perceptions about the costs of different tuition types do not drive the income-contingency effect.

5. Experimental Investigation of Adolescents

While it is the adult population that votes for politicians who implement different policies, chosen payment schemes directly affect future university students. Therefore, the preferences of adolescents may be particularly relevant in shaping higher education policy not only because of their active role as current stakeholders in the education system, future voters, and potential

²⁶ This design intentionally circumvents the necessity of performing discounting calculations, which may be impossible for many respondents and may thus bias response patterns across different participant groups.

protesters against tuition, but also because their decisions whether or not to pursue a higher education may be affected by the tuition design.

5.1 The Adolescent Survey and Experimental Design

To study the effect of tuition payment schemes on the political preferences of adolescents, we conducted a separate representative survey of adolescents. Fielded in 2018 as an online survey, it comprises 1,085 respondents who are representative of the German population aged 14 to 17 years. The recruitment of adolescents was operationalized in two ways. First, 318 adolescents who were registered at online-access panels were recruited directly. Second, 767 adolescents were recruited indirectly via their parents who were registered at online panels. These parents were first asked for their permission to survey their adolescent child, and, if accepted, received a survey link to be shared with their child.²⁷ The adolescent survey contains 28 questions (see also Lergetporer, Werner, and Woessmann, 2021), and median response time was 14 minutes. The adolescent sample is weighted using official statistics on age, gender, region of residence, and municipality size.

There are two randomly chosen experimental groups in the adolescent survey. The control group received the same control-group question on regular tuition as in the adult sample. The treatment group received the question on income-contingent tuition with a slightly simplified wording to ease comprehension, describing income-contingent tuition as follows: "In other countries, there is tuition that has to be paid only after graduation – and only if the former students earn enough money."

Appendix Table A17 presents descriptive statistics and shows that observable characteristics are balanced across experimental groups. Of the adolescents, 57 percent expect to earn a university entrance qualification and 52 percent expect to earn a university degree. These majorities would be affected by higher education finance reforms, making adolescents a highly relevant focus group.

5.2 Results for Adolescents

Adolescents' baseline support for tuition is considerably lower than in the adult population. Table 10 shows that a clear majority of 62.9 percent of adolescents in the control group opposes

²⁷ We implemented plausibility checks of age and birth date to assure that children and not their parents answered the survey. In case of failure to provide consistent answers, respondents were exited from the survey. Reassuringly, the income-contingency effect is large and statistically highly significant among both respondents recruited directly and those recruited via their parents, with (marginally) significantly larger effects in the latter group (see Appendix Table A16). Thus, our main results are not driven by a specific mode of recruitment.

regular tuition. Only 26.6 percent of adolescents favor regular tuition. Consistent with our finding that (potential) students are generally less supportive of tuition (see section 3.5), the low support for tuition in the adolescent control group again suggests the importance of potential students' pecuniary self-interest as a determinant of preferences for tuition.

Intriguingly, changing to an income-contingent payment design has almost the same treatment effect among adolescents as it has among adults. The income-contingency treatment increases support for tuition among adolescents by 14.0 percentage points (compared to 15.4 percentage points in the adult sample in the 2018 wave) and reduces opposition by 16.6 percentage points. This highly significant effect reduces the majority opposing regular tuition to only 46.3 percent opposing income-contingent tuition (40.6 percent in favor; 13.1 percent indifferent). The treatment effect is insensitive to the inclusion of covariates (column 2 and 4) and operates through shifting both strongly and weakly held preferences (see Panel B of Appendix Table A7).²⁸

Subgroup analysis reveals that pecuniary self-interest, clearly visible for regular tuition, does not carry over to income-contingent tuition. Unsurprisingly, adolescents who expect to obtain a university degree are less likely to support tuition (see Appendix Table A18, columns 1 and 2). Interestingly, though, support for income-contingent tuition does not differ significantly across these adolescent subgroups (columns 3 and 4). By implication, estimates of treatment effect heterogeneity indicate that the income-contingency treatment is substantially larger for adolescents who expect to go to university (column 5). Additional treatment heterogeneities include a smaller effect for adolescents whose mother is employed and a larger effect (marginally significant) for adolescents who state that they are strongly interested in politics.

In sum, the overall pattern of preferences for regular versus income-contingent tuition in the adult population is also prevalent among adolescents. While – similar to university students in the adult sample – adolescents are much more opposed to regular tuition, the income-contingency treatment substantially increases their preferences for tuition, eliminating the majority opposition against tuition in the control group. Thus, the design of the tuition payment scheme is a key determinant of tuition preferences among the particularly important focus group of adolescents.

²⁸ The large income-contingency treatment effect on adolescents contrasts with the finding that providing information about the university earnings premium does not affect (potential) university students' support for regular tuition (see Lergetporer and Woessmann, 2023).

6. Conclusions

For more than six decades, economists have advocated deferred income-contingent financing schemes of higher education on efficiency and equity grounds (e.g., Friedman, 1955; Barr et al., 2019): Compared to regular upfront tuition, they do not deter children of credit-constrained families from going to university, act as insurance against income risk, and still ensure that graduates who reap the private financial benefits of their university degree repay part of the public costs of their studies.

In this paper, we add a political-economy dimension to the analysis by demonstrating that income-contingent schemes are strongly favored by the electorate and are therefore politically more feasible. Which payment schemes are implemented by policymakers ultimately depends on the electorate's preferences. We conduct randomized experiments in eight representative German adult surveys and an additional adolescent survey with a total of over 25,000 observations to investigate how the design of the payment scheme shapes the electorate's preferences for tuition.

Our results provide robust evidence that designing tuition as deferred income-contingent rather than regular upfront payments has a profound effect on its public appeal. Public opinion on upfront tuition is divided. Designing tuition instead as deferred payments that depend on future income has a large treatment effect of 18.3 percentage points. Thus, tuition has strong majority support of 62.4 percent if it is charged after graduation and only if future income exceeds a certain threshold. The large income-contingency treatment effect replicates in six representative adult survey waves as well as in an adolescent survey, and it materializes with the same magnitude even in a loan framing that makes loan repayment requirements highly salient and even if we experimentally increase respondents' perceptions about the political consequences attached to their answers.

The mechanisms behind the large treatment effect on public support lie in the very features of the income-contingent tuition scheme emphasized in recent economic research. Our mediation analysis suggests that a substantial part of the overall treatment effect on preferences for tuition can be attributed to the fact that respondents perceive income-contingent tuition as fairer than regular tuition, ensuring that also children from poor families can afford to go to university. Perceptions that the income-contingent payment scheme is less detrimental to (prospective) students also play a role.

When separating the two elements of payment deferral and income thresholds, both have significant and substantial effects. However, the effect of deferred payment alone is nearly as

large as the overall effect of deferred income-contingent tuition, indicating a leading role for the postponement of payment streams until after graduation.

Survey experiments have become a standard approach to study a broad range of economic phenomena (e.g., Haaland, Roth, and Wohlfart, 2023). Despite the inherent trade-offs when choosing a specific research design, we believe that several reasons make survey experiments the most suitable method for investigating how the design of the payment scheme affects the electorate's preferences for tuition. First, they allow for the random assignment of alternative payment schemes between respondents, yielding direct causal estimates of how tuition designs influence public support. Such estimates are hardly possible using non-experimental settings, because exogenous variation in the types of tuition is usually lacking. Second, survey experiments can be implemented in large online surveys which are representative for the votingage population, a central aspect when studying political majorities for different tuition schemes in the electorate. Third, there is compelling evidence that preferences expressed in opinion surveys have real-world political implications. Hypothetical survey experiments have demonstrated external validity with outcomes that closely mirror real-world election behavior, such as voting in referenda (e.g., Hainmueller, Hangartner, and Yamamoto, 2015). Moreover, public-opinion surveys inherently influence the political process as they shape politicians' policy decisions (e.g., Blinder and Krueger, 2004; Hager and Hilbig, 2020).

Many countries around the world struggle with soaring fiscal deficits, not least because of the economic challenges posed by the Covid-19 pandemic. Our results suggest that policymakers may have some room for reforming the financing of higher education – a major public expenditure item – *if* the tuition scheme is designed as deferred income-contingent repayments. The political and public controversies over charging university tuition in the past decades (see Lergetporer and Woessmann, 2023) make Germany particularly relevant for studying public preferences for tuition. Obviously, the generalizability of our findings to other country settings requires further research as they may depend on institutional and political adverse effects on students are potential mechanisms why the income-contingent design is politically more feasible than regular tuition provides some indication how results may generalize to other contexts.

References

- Abraham, Katharine G., Emel Filiz-Ozbay, Erkut Y. Ozbay, Lesley J. Turner (2020). Framing effects, earnings expectations, and the design of student loan repayment schemes. *Journal of Public Economics* 183: 104067.
- Alesina, Alberto, Armando Miano, Stefanie Stantcheva (2023). Immigration and redistribution. *Review of Economic Studies* 90 (1): 1-39.
- Alesina, Alberto, Stefanie Stantcheva, Edoardo Teso (2018). Intergenerational mobility and support for redistribution. *American Economic Review* 108 (2): 521-554.
- Barr, Nicholas (2004). Higher education funding. *Oxford Review of Economic Policy* 20 (2): 264-283.
- Barr, Nicholas, Bruce Chapman, Lorraine Dearden, and Susan Dynarski (2019). The US college loans system: Lessons from Australia and England. *Economics of Education Review* 71: 32-48.
- Bietenbeck, Jan, Andreas Leibing, Jan Marcus, Felix Weinhardt (2023). Tuition fees and educational attainment. *European Economic Review* 154: 104431.
- Blinder, Alan S., Alan B. Krueger (2004). What does the public know about economic policy, and how does it know it? *Brookings Papers on Economic Activity* 2: 187-254.
- Borck, Rainald, Martin Wimbersky (2014). Political economics of higher education finance. *Oxford Economic Papers* 66 (1): 115-139.
- Britton, Jack, Laura van der Erve, Tim Higgins (2019). Income contingent student loan design: Lessons from around the world. *Economics of Education Review* 71: 65-82.
- Bursztyn, Leonardo (2016). Poverty and the political economy of public education spending: Evidence from Brazil. *Journal of the European Economic Association* 14 (5): 1101-1128.
- Carneiro, Pedro, James Heckman (2002). The evidence on credit constraints in post-secondary schooling. *Economic Journal* 112 (482): 705-734.
- Carson, Richard T. (2012). Contingent valuation: A practical alternative when prices aren't available. *Journal of Economic Perspectives* 26 (4): 27-42.
- Chapman, Bruce (2006). Income contingent loans for higher education: International reforms. In *Handbook of the Economics of Education*, edited by Eric A. Hanushek, Finis Welch. Amsterdam: North Holland: 1435-1503.
- Chapman, Bruce (2016). Income contingent loans in higher education financing. *IZA World of Labor* 2016: 227.
- Chapman, Bruce, Timothy Higgins, Joseph E. Stiglitz, eds. (2014). *Income contingent loans: Theory, practice and prospects*. International Economic Association Conference Volume No. 153. Houndmills: Palgrave Macmillan.
- Coffman, Lucas C., Muriel Niederle (2015). Pre-analysis plans have limited upside, especially where replications are feasible. *Journal of Economic Perspectives* 29 (3): 81-98.
- Cruces, Guillermo, Ricardo Perez-Truglia, Martin Tetaz (2013). Biased perceptions of income distribution and preferences for redistribution: Evidence from a survey experiment. *Journal of Public Economics* 98: 100-112.
- De Fraja, Gianni (2001). Education policies: Equity, efficiency and voting equilibrium. *Economic Journal* 111 (471): 104-119.
- De Quidt, Jonathan, Johannes Haushofer, Christopher Roth (2018). Measuring and bounding experimenter demand. *American Economic Review* 108 (11): 3266-3302.

- Diris, Ron, Erwin Ooghe (2018). The economics of financing higher education. *Economic Policy* 33 (94): 265-314.
- Dohmen, Thomas, Armin Falk (2010). You get what you pay for: Incentives and selection in the education system. *Economic Journal* 120 (546): F256-F271.
- Dwenger, Nadja, Johanna Storck, Katharina Wrohlich (2012). Do tuition fees affect the mobility of university applicants? Evidence from a natural experiment. *Economics of Education Review* 31 (1): 155-167.
- Dynarski, Susan (2014). An economist's perspective on student loans in the United States. Brookings Institution Economic Studies Working Paper Series. Washington, DC.
- Epple, Dennis, Richard E. Romano (1996). Ends against the middle: Determinating public service provision when there are private alternatives. *Journal of Public Economics* 62 (3): 297-325.
- Falk, Armin, Anke Becker, Thomas Dohmen, David Huffman, Uwe Sunde (2023). The preference survey module: A validated instrument for measuring risk, time, and social preferences. *Management Science* 69 (4): 1935-1950.
- Falk, Armin, Florian Zimmermann (2013). A taste for consistency and survey response behavior. *CESifo Economic Studies* 59 (1): 181-193.
- Fernandez, Raquel, Richard Rogerson (1995). On the political economy of education subsidies. *Review of Economic Studies* 62 (2): 249-262.
- Field, Erica (2009). Educational debt burden and career choice: Evidence from a financial aid experiment at NYU Law School. *American Economic Journal: Applied Economics* 1 (1): 1-21.
- Findeisen, Sebastian, Dominik Sachs (2016). Education and optimal dynamic taxation: The role of income-contingent student loans. *Journal of Public Economics* 138: 1-21.
- Freundl, Vera, Elisabeth Grewenig, Franziska Kugler, Philipp Lergetporer, Ruth Schüler, Katharina Wedel, Katharina Werner, Olivia Wirth, Ludger Woessmann (2023). The ifo Education Survey 2014-2021: A new dataset on public preferences for education policy in Germany. *Journal of Economics and Statistics* 243 (6): 699-710.
- Friedman, Milton (1955). The role of government in education. In *Economics and the Public Interest*, edited by Robert A. Solo. New Brunswick, NJ: Rutgers Unitersity Press: 123-144.
- Grewenig, Elisabeth, Philipp Lergetporer, Lisa Simon, Katharina Werner, Ludger Woessmann (2023). Can internet surveys represent the entire population? A practitioners' analysis. *European Journal of Political Economy* 78: 102382.
- Grewenig, Elisabeth, Philipp Lergetporer, Katharina Werner, Ludger Woessmann (2022). Incentives, search engines, and the elicitation of subjective beliefs: Evidence from representative online survey experiments. *Journal of Econometrics* 231 (1): 304-326.
- Glomm, Gerhard, B. Ravikumar, Ioana C. Schiopu (2011). The political economy of education funding. In *Handbook of the Economics of Education, Vol. 4*, edited by Eric A. Hanushek, Stephen Machin, Ludger Woessmann. Amsterdam: North Holland: 615-680.
- Grossman, Gene M., Elhanan Helpman (2001). *Special Interest Politics*. Cambridge, MA: MIT Press.
- Haaland, Ingar, Christopher Roth (2020). Labor market concerns and support for immigration. *Journal of Public Economics* 191: 104256.

- Haaland, Ingar, Christopher Roth (2023). Beliefs about racial discrimination and support for pro-black policies. *Review of Economics and Statistics* 105 (1): 40-53.
- Haaland, Ingar, Christopher Roth, Johannes Wohlfart (2023). Designing information provision experiments. *Journal of Economic Literature* 61 (1): 3-40.
- Hager, Anselm, Hanno Hilbig (2020). Does public opinion affect political speech? *American Journal of Political Science* 64 (4): 921-937.
- Hainmueller, Jens, Dominik Hangartner, Teppei Yamamoto (2015). Validating vignette and conjoint survey experiments against real-world behavior. *Proceedings of the Naitonal Academy of Sciences* 112 (8): 2395-2400.
- Hanushek, Eric A., Ludger Woessmann (2008). The role of cognitive skills in economic development. *Journal of Economic Literature* 46 (3): 607-668.
- Heckman, James J., Rodrigo Pinto, Peter Savelyev (2013). Understanding the mechanisms through which an influential early childhood program boosted adult outcomes. *American Economic Review* 103 (6): 2052-2086.
- Heckman, James J., Rodrigo Pinto (2015). Econometric mediation analyses: Identifying the sources of treatment effects from experimentally estimated production technologies with unmeasured and mismeasured inputs. *Econometric Reviews* 34 (1-2): 6-31.
- Jacobs, Bas, Frederick van der Ploeg (2006). Guide to reform of higher education: A European perspective. *Economic Policy* 21 (47): 535-592.
- Kauder, Björn, Niklas Potrafke (2013). Government ideology and tuition fee policy: Evidence from the German states. *CESifo Economic Studies* 59 (4): 628-649.
- Kling, Catherine L., Daniel J. Phaneuf, Jinhua Zhao (2012). From Exxon to BP: Has some number become better than no number? *Journal of Economic Perspectives* 26 (4): 3-26.
- Kuziemko, Ilyana, Michael I. Norton, Emmanuel Saez, Stefanie Stantcheva (2015). How elastic are preferences for redistribution? Evidence from randomized survey experiments. *American Economic Review* 105 (4): 1478-1508.
- Lergetporer, Philipp, Guido Schwerdt, Katharina Werner, Martin R. West, Ludger Woessmann (2018). How information affects support for education spending: Evidence from survey experiments in Germany and the United States. *Journal of Public Economics* 167: 138-157.
- Lergetporer, Philipp, Katharina Werner, Ludger Woessmann (2020). Educational inequality and public policy preferences: Evidence from representative survey experiments. *Journal of Public Economics* 188: 104226.
- Lergetporer, Philipp, Katharina Werner, Ludger Woessmann (2021). Does ignorance of economic returns and costs explain the educational aspiration gap? Representative evidence from adults and adolescents. *Economica* 88 (351): 624-670.
- Lergetporer, Philipp, Ludger Woessmann (2023). Earnings information and public preferences for university tuition: Evidence from representative experiments. *Journal of Public Economics* 226: 104968.
- Lochner, Lance J., Alexander Monge-Naranjo (2011). The Nature of Credit Constraints and Human Capital. *American Economic Review* 101 (6): 2487-2529.
- Loewenstein, George, Richard Thaler (1989). Anomalies: Intertemporal choice. *Journal of Economic Perspectives* 3 (4): 181-193.
- Maniadis, Zacharias, Fabio Tufano, John A. List (2014). One swallow doesn't make a summer: New evidence on anchoring effects. *American Economic Review* 104 (1): 277-290.

- Mummolo, Jonathan, Erik Peterson (2019). Demand effects in survey experiments: An empirical assessment. *American Political Science Review* 113 (2): 517-529.
- Murphy, Richard, Judith Scott-Clayton, Gill Wyness (2019). The end of free college in England: Implications for enrolments, equity, and quality. *Economics of Education Review* 71: 7-22.
- OECD (2022). Education at a glance 2022: OECD indicators. Paris: Organisation for Economic Co-operation and Development.
- Oosterbeek, Hessel, Anja van den Broek (2009). An empirical analysis of borrowing behaviour of higher education students in the Netherlands. *Economics of Education Review* 28 (2): 170-177.
- Roth, Christopher, Sonja Settele, Johannes Wohlfart (2022). Beliefs about public debt and the demand for government spending. *Journal of Econometrics* 231 (1): 165-187.
- Settele, Sonja (2022). How do beliefs about the gender wage gap affect the demand for public policy? *American Economic Journal: Economic Policy* 14 (2): 475-508.
- Stantcheva, Stefanie (2017). Optimal taxation and human capital policies over the life cycle. *Journal of Political Economy* 125 (6): 1931-1990.
- Woessmann, Ludger, Philipp Lergetporer, Franziska Kugler, Katharina Werner (2014). Was die Deutschen über die Bildungspolitik denken: Ergebnisse des ersten ifo Bildungsbarometers. *ifo Schnelldienst* 67 (18): 16–33.





■ strongly favor ■ somewhat favor ■ strongly oppose ■ somewhat oppose

Notes: Control group: preferences for regular upfront tuition. Income-contingency treatment group: preferences for deferred income-contingent tuition (scheme of deferred payments charged only if future income exceeds a threshold). Residual category: "neither favor nor oppose." Data source: ifo Education Survey 2015, 2016, and 2018-2020.

Table 1: Data structure

Analysis and wave	Control group (regular tuition)	Treatment group (income contingency)	Total
	(1)	(2)	(3)
Baseline experiment			
2015	1,390	1,360	2,750
2016	781	852	1,633
2018	1,036	1,005	2,041
2019 (incl. mechanism experiment)	2,013	1,996	4,009
2020	1,050	1,055	2,105
2023	1,095 (incl. within-subject design)	1,151	2,246
Total baseline experiment	6,270	6,268	14,784
Within-subject design – 2014	1,032		
2017	2,075		
Total	3,107		3,107
<i>Loan-framing experiment</i> – 2018		Loan framing: 970	970
Consequentiality experiment – 2020	Consequential: 1,024	Consequential: 1,058	2,082
Separate elements experiment – 2023		Payment deferral alone: 1,139	1,139
		Income threshold alone: 1,137	1,137
		Without "other countries": 1,114	1,114
Total adult sample			24,333
Experiment in adolescent survey – 2018	525	560	1,085
Total sample			25,418

Notes: Numbers of observations per experimental group. Numbers include participants with missing responses on the tuition question. The total number of responses on the tuition question is 25,364 observations (24,279 adult observations and 1,085 adolescent observations). See Appendix Table A2 for item non-response. Data source: ifo Education Survey 2014-2020, 2023.

	Support	for tuition	Opposition a	gainst tuition
	(1)	(2)	(3)	(4)
Income contingency	0.153 ^{***}	0.149***	-0.158***	-0.153***
	(0.026)	(0.026)	(0.026)	(0.025)
Income contingency \times Wave 2016	-0.000	0.004	-0.007	-0.013
	(0.040)	(0.039)	(0.039)	(0.038)
Income contingency \times Wave 2018	0.001	0.011	-0.033	-0.043
	(0.042)	(0.041)	(0.040)	(0.040)
Income contingency \times Wave 2019	0.062^{*}	0.068 ^{**}	-0.049	-0.059*
	(0.034)	(0.033)	(0.032)	(0.032)
Income contingency \times Wave 2020	0.068 [*]	0.071 ^{**}	-0.048	-0.054
	(0.035)	(0.034)	(0.034)	(0.033)
Wave 2016	-0.014	-0.008	-0.003	0.001
	(0.028)	(0.028)	(0.028)	(0.028)
Wave 2018	0.033	0.021	-0.011	0.003
	(0.029)	(0.029)	(0.029)	(0.029)
Wave 2019	0.007	0.000	-0.037	-0.026
	(0.024)	(0.024)	(0.024)	(0.024)
Wave 2020	-0.026	-0.028	0.008	0.014
	(0.025)	(0.025)	(0.025)	(0.025)
Covariates	No	Yes	No	Yes
Control mean	0.438	0.438	0.464	0.464
Observations R^2	12,504	12,504	12,504	12,504
	0.036	0.059	0.042	0.066

Table 2: The effect of deferred income-contingent payment schemes on preferences for tuition

Notes: OLS regressions. *Income contingency*: experimental treatment in the survey experiment. Sample: control groups and income-contingency treatment groups in waves 2015, 2016, and 2018-2020. Dependent variable: col. (1)-(2): dummy variables 1 = "strongly favor" or "somewhat favor" tuition, 0 otherwise; col. (3)-(4): dummy variables 1 = "strongly oppose" or "somewhat oppose" tuition, 0 otherwise. Residual category: "neither favor nor oppose." Control mean: mean of the outcome variable in the control group in wave 2015. Covariates include age, income, and dummies for gender, born in Germany, living in a tuition state, living with partner in household, parents' university degree, working in the education sector, highest educational attainment, professional degree, employment status, parent status, political party preferences, and voting behavior. Missing values of covariates are imputed; regressions with covariates include imputation dummies. Data source: ifo Education Survey 2015-2020. Regressions weighted by survey weights. Robust standard errors in parentheses. Significance levels: *** p<0.01, ** p<0.05, * p<0.10.

Table 2.	Within aul	icot com	nomicon of	f nnofonon and	for regular and	income contingent twitien
I able 5:	vvilnin-sui	несь сот	darison oi	Dreferences	tor regular and	I Income-contingent tuition

	Support	Neutral	Oppose	Total
Regular tuition	(1)	(2)	(3)	(4)
Support	0.334	0.024	0.057	0.415
Neutral	0.057	0.071	0.011	0.139
Oppose	0.252	0.048	0.147	0.447
Total	0.642	0.143	0.215	1.000

Notes: Sample: control groups in waves 2014 and 2017. Shares of respondents who support, oppose, or are neutral towards tuition (weighted group means). Data source: ifo Education Survey 2014, 2017.

	Support for tuition		Opposition against tuition		
	(1)	(2)	(3)	(4)	
Income contingency	0.154***	0.160***	-0.191***	-0.191***	
	(0.032)	(0.032)	(0.031)	(0.030)	
Loan framing	0.150***	0.146***	-0.197***	-0.192***	
	(0.032)	(0.031)	(0.030)	(0.029)	
Covariates	No	Yes	No	Yes	
Control mean	0.472	0.472	0.453	0.453	
Observations	3,011	3,011	3,011	3,011	
R^2	0.022	0.059	0.039	0.066	

Table 4: The effect of loan framing on preferences for income-contingent tuition

Notes: OLS regressions. *Income contingency* and *Loan framing*: experimental treatments in the survey experiment. Sample: wave 2018. Dependent variable: col. (1)-(2): dummy variables 1 = "strongly favor" or "somewhat favor" tuition, 0 otherwise; col. (3)-(4): dummy variables 1 = "strongly oppose" or "somewhat oppose" tuition, 0 otherwise. Residual category: "neither favor nor oppose." Control mean: mean of the outcome variable in the control group. Covariates include age, income, and dummies for gender, born in Germany, living in a tuition state, city size, living with partner in household, parents' university degree, working in the education sector, highest educational attainment, professional degree, employment status, parent status, political party preferences, and voting behavior. Missing values of covariates are imputed; regressions with covariates include imputation dummies. Data source: ifo Education Survey 2018. Regressions weighted by survey weights. Robust standard errors in parentheses. Significance levels: *** p<0.01, ** p<0.05, * p<0.10.

	Main experiment (ifo Education Survey 2020)				Supplemental experiment (supplemental survey 2023)				
	Support for tuition		Oppo against	Opposition against tuition		Support for income- contingent tuition		Perceived consequentiality of survey response	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
Perceived consequentiality	0.029 (0.024)	0.033 (0.023)	-0.002 (0.024)	-0.005 (0.023)	0.006 (0.021)	0.008 (0.020)	0.090^{***} (0.021)	0.091*** (0.021)	
Income contingency	0.221*** (0.023)	0.221*** (0.023)	-0.206*** (0.022)	-0.206 ^{***} (0.022)					
Income conti. + Perceived conse.	0.205 ^{***} (0.023)	0.211 ^{***} (0.023)	-0.184 ^{***} (0.022)	-0.191 ^{***} (0.022)					
Covariates	No	Yes	No	Yes	No	Yes	No	Yes	
Control mean	0.412	0.412	0.472	0.472	0.631	0.631	0.540	0.540	
Observations R^2	4,187 0.040	4,187 0.078	4,187 0.041	4,187 0.085	2,196 0.000	2,196 0.036	2,196 0.008	2,196 0.016	

Table 5.	Inorposing	norcontions	about	nalition	aancaa	montiality	of curve	w roonongog
Table 5.	mereasing	perceptions	avout	ponucai	conseq	uentianty	y of surve	y responses

Notes: OLS regressions. *Perceived consequentiality, Income contingency*, and *Income contingency* + *Perceived consequentiality*: experimental treatments in the survey experiment. Samples: wave 2020, supplemental survey 2023. Dependent variable: col. (1)-(2), (5)-(6): dummy variables 1 = "strongly favor" or "somewhat favor" tuition, 0 otherwise; col. (3)-(4): dummy variables 1 = "strongly oppose" or "somewhat oppose" tuition, 0 otherwise; col (7)-(8): dummy variable 1 = "fully agree" or "rather agree" that respondents' answers can impact education policy in their state, 0 otherwise. Control mean: mean of the outcome variable in the control group. Covariates include age, income, and dummies for gender, living in a tuition state, parents' university degree, highest educational attainment, professional degree, parent status, and voting behavior; additional covariates in col. (1)-(4): born in Germany, living with partner in household, working in the education Survey 2020 and supplemental survey 2023. Regressions (1)-(4) weighted by survey weights. Robust standard errors in parentheses. Significance levels: *** p<0.01, ** p<0.05, * p<0.10.

Treatment (omitted subgroup) (1)		Treatment-subgroup interaction (2)				
Sociodemographic characteristics						
Age 45-65	0.217***	\times Age under 45	-0.090***			
	(0.015)		(0.024)			
		\times Age over 65	0.009			
			(0.031)			
Male	0.159***	×Female	0.045^{**}			
	(0.017)		(0.023)			
Born in Germany	0.166***	\times Not born in Germany	0.017			
	(0.052)		(0.053)			
No tuition state	0.213***	\times Tuition state	-0.044*			
	(0.020)		(0.024)			
Monthly household income below median	0.170^{***}	\times Income above median	0.026			
	(0.016)		(0.023)			
No partner in household	0.170^{***}	\times Partner in household	0.022			
	(0.019)		(0.024)			
Has parent(s) without university degree	0.188^{***}	\times Has parent(s) with university degree	-0.020			
	(0.014)		(0.025)			
Not working in education sector	0.187^{***}	\times Works in education sector	-0.051			
	(0.012)		(0.038)			
Not full-time employed	0.190^{***}	\times Full-time employed	-0.020			
	(0.015)		(0.023)			
Political activity and preferences						
Frequent voter	0.194***	\times No frequent voter	-0.066**			
	(0.013)		(0.030)			
Education important for vote choice	0.197***	\times Education not important for vote choice	-0.080***			
L	(0.015)	ľ	(0.029)			
No CDU partisan	0.179***	\times CDU partisan	0.009			
L	(0.013)	1	(0.028)			
Interest groups						
No school degree	0 149***	× Middle school degree	0.036			
	(0.020)	× mildle senoor degree	(0.028)			
	(0.020)	× University entrance qualification	0.071***			
		× oniversity enhance quanteation	(0.071)			
No university degree	0 178***	× University degree	0.024			
ito university degree	(0.013)		(0.029)			
No university student	0.183***	× University student	-0.035			
No university student	(0.103)	~ Oniversity student	-0.033			
No children	0.146***	× At least one child < 18	0.018			
i to emilaten	(0,010)	A rational one ennu × 10	(0.020)			
	(0.01))	\times All children > 18	0.091***			
			(0.027)			

Table 6: Subgroup analysis

Notes: Each row represents a separate OLS regression according to equation (2) that includes the experimental income-contingency treatment and its interaction term(s) with subgroup indicator(s). Dependent variable: support for tuition (dummy variable 1 = "strongly favor" or "somewhat favor" tuition, 0 otherwise). Col. (1): coefficient on *Income contingency*, reflecting the treatment effect in the respective omitted subgroup. Col. (2): coefficient(s) on the interaction term(s) between *Income contingency* and the indicated subgroup indicator(s), reflecting the difference in the treatment effect between the omitted subgroup and the respective indicated subgroup. Data source: ifo Education Survey 2015-2020. Regressions weighted by survey weights. Robust standard errors (clustered at the individual level) in parentheses. Significance levels: *** p<0.01, ** p<0.05, * p<0.10.

	Agreement to statements relating to									
	f	fairness conc	erns	(pr	ospective) stu	udents	gove	government and universities		
	Despite tuition, poor Tuition can afford is unjust to study		Tuition makes high earners contribute	Tuition deters people from studying	TuitionTuitiondetersrestrictspeople fromchoice ofstudyingsubject		Tuition is hard to administer	Tuition relieves state finance	Tuition improves university quality	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	
Income contingency	0.322***	-0.186***	0.062**	-0.153***	-0.162***	-0.141***	0.017	-0.005	0.078***	
	(0.027)	(0.026)	(0.025)	(0.029)	(0.028)	(0.029)	(0.026)	(0.028)	(0.027)	
Covariates	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	
Control mean	0.330	0.418	0.704	0.532	0.508	0.595	0.265	0.658	0.609	
Observations	1,991	1,991	1,991	1,991	1,991	1,991	1,991	1,991	1,991	
R^2	0.148	0.075	0.061	0.061	0.089	0.057	0.049	0.031	0.051	

Table 7: Potential mechanisms: Effects of income-contingent payment schemes on perceptions about tuition

Notes: OLS regressions. *Income contingency*: experimental treatment in the survey experiment. Sample: control group and income-contingency treatment group in wave 2019 (randomly chosen subsamples). Dependent variable: dummy variables 1 = respective statement is "fully true" or "rather true", 0 otherwise. Control mean: mean of the outcome variable in the control group. Covariates include age, income, patience, risk tolerance, and dummies for gender, born in Germany, living in a tuition state, city size, living with partner in household, parents' university degree, working in the education sector, highest educational attainment, professional degree, employment status, parent status, political party preferences, and voting behavior. Missing values of covariates are imputed; regressions with covariates include imputation dummies. Data source: ifo Education Survey 2019. Regressions weighted by survey weights. Robust standard errors in parentheses. Significance levels: *** p<0.01, ** p<0.05, * p<0.10.

		Share of treatment effect attributed to								
	fa	airness conc	cerns	(pr	ospective) stu	idents	universities			
	Despite tuition, poor can afford to study	Tuition is unjust	Tuition makes high earners contribute	Tuition deters people from studying	Tuition restricts choice of subject	Tuition increases debt risk	Tuition improves university quality	all channels jointly		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)		
Panel A: Entered separately										
Share attributed to observed factor	0.360	0.218	0.080	0.127	0.137	0.136	0.080			
Panel B: Entered jointly										
Share attributed to observed factor	0.191	0.106	0.052	0.021	0.030	0.067	0.043			
		0.349			0.117		0.043	0.509		
Share left to unobserved factors								0.491		

Table 8: Mediation analysis: Share of income-contingency treatment effect attributed to mediators

Notes: Share of the income-contingency treatment attributed to the respective mediator (using the mediation approach of Heckman, Pinto, and Savelyev, 2013). Panel A: contributions of the different mediators considered separately; Panel B: contributions of the different mediators considered jointly. Data source: ifo Education Survey 2019.

	Support for tuition		Opposition a	gainst tuition
	(1)	(2)	(3)	(4)
Income contingency	0.138^{***}	0.138^{***}	-0.175^{***}	-0.169^{***}
Payment deferral alone (no income threshold)	(0.02.1) 0.112^{***} (0.034)	(0.000) 0.106^{***} (0.033)	-0.169^{***}	-0.161^{***}
Income threshold alone (no payment deferral)	0.073**	0.069** (0.034)	-0.118^{***} (0.033)	-0.110^{***} (0.032)
Income contingency without mentioning "other countries"	0.126 ^{***} (0.034)	0.123*** (0.034)	-0.148*** (0.032)	-0.141*** (0.031)
Covariates	No	Yes	No	Yes
Difference between Income contingency and				
Payment deferral alone (no income threshold)	0.026	0.032	-0.007	-0.008
Income threshold alone (no payment deferral)	0.065^*	0.069^{**}	-0.058^{*}	-0.059**
Income cont. without mentioning "other countries"	0.012	0.015	-0.027	-0.028
Control mean	0.460	0.460	0.423	0.423
Observations	5,632	5,632	5,632	5,632
R^2	0.010	0.036	0.019	0.047

Table 9: Payment deferral and income threshold as separate elements (and lack of reference to other countries)

Notes: OLS regressions. *Income contingency, Payment deferral alone (no income threshold), Income threshold alone (no payment deferral)*, and *Income contingency without mentioning "other countries"*: experimental treatments in the survey experiment. Sample: wave 2023. Dependent variable: col. (1)-(2): dummy variables 1 = "strongly favor" or "somewhat favor" tuition, 0 otherwise; col. (3)-(4): dummy variables 1 = "strongly oppose" or "somewhat oppose" tuition, 0 otherwise. Residual category: "neither favor nor oppose." Difference between Income contingency and other treatments: Coefficient on Income contingency minus coefficient on the respective other treatment group; significance of differences calculated using post-estimation Wald tests. Control mean: mean of the outcome variable in the control group. Covariates include age, income, and dummies for gender, born in Germany, living in a tuition state, city size, living with partner in household, parents' university degree, highest educational attainment, professional degree, employment status, parent status, political party preferences, and voting behavior. Missing values of covariates are imputed; regressions with covariates include imputation dummies. Data source: ifo Education Survey 2023. Regressions weighted by survey weights. Robust standard errors in parentheses. Significance levels: *** p<0.01, ** p<0.05, * p<0.10.

	Support f	for tuition	Opposition against tuition		
	(1)	(2)	(3)	(4)	
Income contingency	0.140 ^{***} (0.030)	0.138 ^{***} (0.030)	-0.166*** (0.032)	-0.167*** (0.031)	
Covariates	No	Yes	No	Yes	
Control mean	0.266	0.266	0.629	0.629	
Observations R^2	1,085 0.022	1,085 0.049	1,085 0.028	1,085 0.067	

Table 10: Effects of income-contingent payment schemes on preferences for tuition among adolescents

Notes: OLS regressions. *Income contingency*: experimental treatment in the survey experiment. Sample: adolescent survey, 2018. Dependent variable: col. (1)-(2): dummy variables 1 = "strongly favor" or "somewhat favor" tuition, 0 otherwise; col. (3)-(4): dummy variables 1 = "strongly oppose" or "somewhat oppose" tuition, 0 otherwise. Residual category: "neither favor nor oppose." Control mean: mean of the outcome variable in the control group. Covariates include age, patience, risk tolerance, and dummies for gender, born in Germany, living in a tuition state, city size, living with both parents in household, parents' university degree, expected educational attainment, expected professional degree, parents' employment status, having siblings (who study/studied), and political interest. Missing values of covariates are imputed; regressions with covariates include imputation dummies. Data source: ifo Education Survey 2018 (adolescent survey). Robust standard errors in parentheses. Significance levels: *** p<0.01, ** p<0.05, * p<0.10.

German original Item English translation Control group Do you favor or oppose that students at German universities Sind Sie dafür oder dagegen, dass Studierende, die an einer Universität oder Hochschule^a cover a part of the costs of their studies themselves by tuition? in Deutschland studieren, einen Teil der Studienkosten durch Studiengebühren tragen? Income In other countries, there is tuition that is due only after In anderen Ländern gibt es Studiengebühren, die erst nach dem Abschluss des Studiums erhoben werden, wenn die ehemaligen Studierenden Einkommen erzielen. Die Gebühren graduation, when the former students earn income. Tuition has contingency to be paid only if their annual income exceeds a certain müssen nur dann zurückgezahlt werden, wenn ihr Jahreseinkommen über einem gewissen threshold. Do you favor or oppose that students at German Schwellenwert liegt. Sind Sie dafür oder dagegen, dass Studierende, die an einer universities cover a part of the costs of their studies themselves Universität oder Hochschule in Deutschland studieren, einen Teil der Studienkosten by this alternative form of tuition? durch eine solche Form von Studiengebühren tragen? Loan framing In other countries, there is tuition that is being paid in the form In anderen Ländern gibt es Studiengebühren, die in Form von zinsfreien Studienkrediten of interest-free student loans. Loan repayments are due only after bezahlt werden. Die Kreditrückzahlungen sind erst nach dem Abschluss des Studiums graduation, when the former students earn income. The student fällig, wenn die ehemaligen Studierenden Einkommen erzielen. Die Studienkredite loans only have to be paid if their annual income exceeds a müssen nur dann zurückgezahlt werden, wenn ihr Jahreseinkommen über einem gewissen certain threshold. Do you favor or oppose that students at Schwellenwert liegt. Sind Sie dafür oder dagegen, dass Studierende, die an einer German universities cover a part of the costs of their studies Universität oder Hochschule in Deutschland studieren, einen Teil der Studienkosten themselves by this alternative form of tuition? durch diese alternative Form von Studiengebühren tragen? Consequential Information: The average answers to the following question will Information: Die durchschnittlichen Antworten auf die nachfolgende Frage werden nach Abschluss der Befragung an die Landtagspolitiker*innen Ihres Bundeslandes be passed on to the state parliament politicians of your federal state after the survey has been completed. Your answer is weitergegeben. Ihre Antwort ist daher besonders wichtig. ... therefore particularly important. ... Mechanisms In your view, to what extent do the following statements about Was denken Sie, inwieweit treffen folgende Aussagen zu Studiengebühren zu? Denken Sie dabei an reguläre Studiengebühren, die zu Beginn jedes Semesters (Halbjahrs) tuition apply? Think of regular tuition that is charged at the beginning of each semester (half year). / ... of income-contingent erhoben werden. / ... an nachgelagerte Studiengebühren, die erst nach dem Abschluss des tuition that is due only after graduation and only if the annual Studiums erhoben werden und nur, wenn das Jahreseinkommen der ehemaligen income of the former students exceeds a certain threshold. Studierenden über einem gewissen Schwellenwert liegt. - Durch die Gebühren wird der Staat finanziell entlastet. - Through tuition, the state is relieved of financial burden. - Through tuition, future high earners contribute to the cost of - Durch die Gebühren tragen zukünftige Besserverdienende zu den Kosten des Studiums bei. studying. - Die Gebühreneinnahmen ermöglichen den Universitäten, ihre Qualität zu verbessern. - Tuition income enables universities to improve their quality. - Auch Personen aus ärmeren Familien können sich trotz Gebühren ein Studium leisten. - Also people from poorer families can afford to study despite tuition. - Die Gebühren halten Personen vom Studieren ab. - Tuition deters people from studying. - Die Gebühren sind ungerecht. - Tuition is unjust. - Die Gebühren erhöhen das Risiko, dass sich Studierende verschulden. - Tuition increases the risk of students getting into debt. - Die Gebühren schränken die Studienfach- und Berufswahl der Studierenden ein. - Tuition restricts students' choice of subject and profession. - Die Gebühren sind für den Staat schwer zu verwalten. - Tuition is difficult for the state to administer.

Table A1: Wording of survey items

(continued on next page)

Table A1 (continued)

Item	English translation	German original
Level	How high do you think tuition should be at most? Think of regular tuition that is charged at the beginning of each semester (half year). / of income-contingent tuition that is due only after graduation and only if the annual income of the former students exceeds a certain threshold.	Wie hoch sollten Studiengebühren Ihrer Meinung nach höchstens sein? Denken Sie dabei an reguläre Studiengebühren, die zu Beginn jedes Semesters (Halbjahrs) erhoben werden. / an nachgelagerte Studiengebühren, die erst nach dem Abschluss des Studiums erhoben werden und nur, wenn das Jahreseinkommen der ehemaligen Studierenden über einem gewissen Schwellenwert liegt.
Payment deferral alone (no income threshold)	There is the possibility of tuition that is due only after graduation, when the former students earn income. Do you favor or oppose that students at German universities cover a part of the costs of their studies themselves by this alternative form of tuition?	Es gibt die Möglichkeit von Studiengebühren, die erst nach dem Abschluss des Studiums erhoben werden, wenn die ehemaligen Studierenden Einkommen erzielen. Sind Sie dafür oder dagegen, dass Studierende, die an einer Universität oder Hochschule in Deutschland studieren, einen Teil der Studienkosten durch diese alternative Form von Studiengebühren tragen?
Income threshold alone (no payment deferral)	There is the possibility of tuition that is due only if the annual income of students' parents exceeds a certain threshold. Do you favor or oppose that students at German universities cover a part of the costs of their studies themselves by this alternative form of tuition?	Es gibt die Möglichkeit von Studiengebühren, die nur dann gezahlt werden müssen, wenn das Jahreseinkommen der Eltern der Studierenden über einem gewissen Schwellenwert liegt. Sind Sie dafür oder dagegen, dass Studierende, die an einer Universität oder Hochschule in Deutschland studieren, einen Teil der Studienkosten durch diese alternative Form von Studiengebühren tragen?
Income contingency without mentioning "other countries"	There is the possibility of tuition that is due only after graduation, when the former students earn income. Tuition has to be paid only if their annual income exceeds a certain threshold. Do you favor or oppose that students at German universities cover a part of the costs of their studies themselves by this alternative form of tuition?	Es gibt die Möglichkeit von Studiengebühren, die erst nach dem Abschluss des Studiums erhoben werden, wenn die ehemaligen Studierenden Einkommen erzielen. Die Gebühren müssen nur dann zurückgezahlt werden, wenn ihr Jahreseinkommen über einem gewissen Schwellenwert liegt. Sind Sie dafür oder dagegen, dass Studierende, die an einer Universität oder Hochschule in Deutschland studieren, einen Teil der Studienkosten durch diese alternative Form von Studiengebühren tragen?

Notes: Answer categories: strongly favor; somewhat favor; somewhat oppose; strongly oppose; neither favor nor oppose (Ich bin sehr dafür; eher dafür; eher dagegen; sehr dagegen; weder dafür noch dagegen). Answer categories for mechanisms: fully true; rather true; rather not true; not true at all; neither nor (Trifft voll zu; trifft eher zu; trifft eher zu; trifft eher zu; trifft überhaupt nicht zu; weder noch). Levels were elicited using an open numerical field: ... Euro per semester (half-year) (... Euro pro Semester (Halbjahr)). ^a Literal translation: "universities or universities of applied sciences."

	Mean in control group					Difference to treatment group									
	2014	2015	2016	2017	2018	2019	2020	2015	2016	2018	2018 ^a	2019	2020	2020 ^b	2020 ^c
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)
Age	51.00	49.40	50.11	49.57	51.51	50.77	44.08	1.68	2.44^{*}	-1.39	-2.18	-0.74	0.74	0.19	0.83
Female	0.50	0.53	0.51	0.51	0.51	0.51	0.49	-0.02	0.02	-0.00	-0.02	-0.01	0.00	0.04^{*}	-0.01
Born in Germany	0.94	0.95	0.96	0.95	0.95	0.95	0.95	-0.01	-0.00	0.02^{*}	0.00	-0.01	-0.02	0.00	-0.01
Tuition state	0.69	0.73	0.72	0.70	0.72	0.72	0.69	-0.06	-0.01	-0.03	0.00	-0.03	0.01	0.04^{*}	0.04^{*}
City size $\geq 100,000$	0.33	0.32	0.31	0.32	0.32	0.33		0.00	-0.01	-0.01	-0.00	-0.02			
Monthly household income (ϵ)	2,133	2,215	2,131	2,252	2,554	2,557	2,710	50.3	41.9	-189.8**	-140.2	-2.2	-49.9	28.9	151.8^{*}
Partner in household	0.58	0.59	0.54	0.55	0.60	0.62	0.59	0.03	0.01	-0.03	-0.03	-0.05**	0.01	0.03	0.02
Parent(s) with univ. degree	0.24	0.27	0.23	0.28	0.30	0.26	0.31	0.00	0.01	-0.03	-0.02	0.03	0.01	0.00	0.03
Works in education sector	0.06	0.11	0.08	0.08	0.10	0.11	0.13	0.00	-0.00	0.00	0.00	0.01	-0.00	-0.01	0.01
Highest education attainment															
No degree/basic degree	0.40	0.39	0.38	0.37	0.37	0.40	0.32	0.03	0.02	-0.00	-0.00	-0.09***	0.01	0.01	-0.02
Middle school degree	0.32	0.31	0.32	0.31	0.31	0.28	0.29	-0.01	-0.02	0.00	-0.03	0.05^{***}	-0.00	-0.01	-0.01
University entrance qualif.	0.28	0.30	0.30	0.32	0.32	0.32	0.39	-0.02	-0.01	0.00	0.03	0.04^{*}	-0.00	0.00	0.03
Professional degree															
No degree	0.10	0.13	0.14	0.09	0.06	0.09	0.10	-0.03*	-0.01	0.02	0.02	-0.01	0.01	-0.02	-0.01
Vocational degree	0.72	0.70	0.70	0.71	0.75	0.71	0.66	0.02	-0.01	-0.03	-0.01	-0.02	0.00	0.03	0.00
University degree	0.15	0.16	0.12	0.15	0.17	0.18	0.22	0.03	0.01	-0.01	0.01	0.01	-0.00	0.00	0.02
In training	0.07	0.06	0.06	0.09	0.06	0.05	0.05	-0.02	0.01	0.02	-0.02	0.02	0.00	0.00	0.01
University student	0.05	0.05	0.05	0.06	0.03	0.03	0.03	-0.03**	0.00	0.02^*	0.00	0.01	0.00	0.00	0.01
Employment status															
Full-time employed	0.37	0.33	0.34	0.37	0.38	0.40	0.50	0.05^{**}	-0.02	0.01	0.01	0.00	0.00	-0.02	-0.01
Part-time employed	0.09	0.13	0.15	0.12	0.13	0.15	0.15	0.00	-0.00	0.03	0.03	-0.00	0.00	0.00	0.01
Self-employed	0.02	0.04	0.03	0.04	0.05	0.04	0.06	-0.01	-0.00	-0.01	0.01	0.02	-0.02	-0.03**	-0.02
Unemployed	0.07	0.07	0.06	0.05	0.03	0.03	0.04	-0.03**	0.00	0.00	0.01	0.00	0.01	0.02	-0.00

 Table A2: Summary statistics and balancing tests: Waves 2015-2020

(continued on next page)

Table A2 (continued)

			Mean i	n contro	l group					Differ	ence to	treatment	group		
	2014	2015	2016	2017	2018	2019	2020	2015	2016	2018	2018 ^a	2019	2020	2020 ^b	2020 ^c
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)
Parent status															
No children	0.37	0.32	0.43	0.43	0.40	0.39	0.46	-0.00	-0.02	-0.00	0.00	0.05**	-0.02	-0.02	-0.01
At least one child < 18	0.23	0.31	0.20	0.208	0.23	0.21	0.25	-0.02	0.00	-0.03	-0.00	-0.01	-0.01	0.00	-0.03
All children > 18	0.40	0.37	0.37	0.37	0.37	0.40	0.29	0.02	0.02	0.04	0.00	-0.03	0.03	0.02	0.03
Political party preference															
CDU	0.26	0.26	0.21	0.25	0.21	0.19	0.22	-0.05**	0.00	-0.01	0.02	0.02	0.00	0.02	-0.02
SPD	0.22	0.22	0.17	0.20	0.20	0.18	0.12	0.02	0.02	-0.02	-0.02	-0.03**	0.02	-0.00	0.01
Linke	0.05	0.08	0.08	0.08	0.11	0.08	0.09	0.01	-0.01	-0.03	0.00	0.00	0.01	-0.01	-0.01
Grüne	0.09	0.07	0.08	0.06	0.07	0.11	0.13	0.02	-0.01	-0.01	0.01	0.01	-0.01	-0.00	-0.01
Other	0.06	0.05	0.15	0.14	0.18	0.20	0.17	-0.00	-0.01	0.02	-0.01	0.01	-0.02	0.01	0.01
None	0.33	0.33	0.32	0.29	0.23	0.24	0.29	0.01	0.01	0.04	0.01	-0.00	-0.01	-0.02	0.02
Patience (11-point scale)	6.86		5.79	6.04	6.15	6.40	7.40		0.18	0.17	0.13	0.03	-0.05	-0.04	-0.10
Altruism (11-point scale)	7.15														
Risk tolerance (11-point scale)			4.24	4.27	4.32	4.78	5.43		0.00	-0.12	0.23	0.06	-0.14	-0.23*	0.04
Frequent voter		0.77	0.77	0.82	0.85	0.85	0.82	0.02	0.03	-0.03	-0.00	0.01	-0.00	-0.00	-0.02
Education important vote		0.72	0.75	0.74	0.75	0.71		0.02	0.03	-0.04	-0.04	0.02			
Non-response	0.01	0.01	0.01	0.01	0.00	0.00	0.00	0.01	0.00	0.00	0.00	0.00	0.00	0.00	0.00
Observations	1,032	1,390	781	2,075	1,036	2,013	1,050	1,360	852	1,005	970	1,996	1,055	1,024	1,058

Notes: Weighted group means. "Difference" displays the difference in means between the control group of the respective year and the treatment groups. Significance levels of "difference" stem from linear regressions of the respective background variable on treatment dummies. Data source: ifo Education Survey 2014-2020. Regressions weighted by survey weights. Significance levels: *** p<0.01, ** p<0.05, * p<0.10. a Loan framing. b Consequential. c Income contingency + Consequential.

	Support	for tuition	Opposition a	gainst tuition
	(1)	(2)	(3)	(4)
Income contingency	0.169***	0.167***	-0.187***	-0.183***
	(0.019)	(0.019)	(0.018)	(0.018)
Income contingency × Wave 2016	0.014	0.017	0.002	-0.002
	(0.031)	(0.030)	(0.030)	(0.029)
Income contingency \times Wave 2018	0.011	0.012	-0.014	-0.016
	(0.029)	(0.029)	(0.028)	(0.027)
Income contingency × Wave 2019	0.058^{**}	0.061**	-0.043*	-0.049**
	(0.024)	(0.024)	(0.023)	(0.023)
Income contingency \times Wave 2020	0.056^{**}	0.058^{**}	-0.028	-0.033
	(0.028)	(0.028)	(0.027)	(0.027)
Wave 2016	-0.005	0.001	-0.008	-0.007
	(0.022)	(0.022)	(0.022)	(0.022)
Wave 2018	0.008	0.009	0.007	0.005
	(0.020)	(0.020)	(0.020)	(0.020)
Wave 2019	-0.006	0.005	-0.006	-0.017
	(0.017)	(0.018)	(0.017)	(0.018)
Wave 2020	-0.032	-0.020	0.021	0.015
	(0.020)	(0.021)	(0.020)	(0.021)
Covariates	No	Yes	No	Yes
Control mean	0.435	0.435	0.464	0.464
Observations	12,504	12,504	12,504	12,504
R^2	0.041	0.062	0.048	0.074

Table A3: Baseline results: Specification without weights

Notes: OLS regressions. *Income contingency*: experimental treatment in the survey experiment. Sample: control groups and income-contingency treatment groups in waves 2015, 2016, and 2018-2020. Dependent variable: col. (1)-(2): dummy variables 1 = "strongly favor" or "somewhat favor" tuition, 0 otherwise; col. (3)-(4): dummy variables 1 = "strongly oppose" or "somewhat oppose" tuition, 0 otherwise. Residual category: "neither favor nor oppose." Control mean: mean of the outcome variable in the control group in wave 2015. Covariates include age, income, and dummies for gender, born in Germany, living in a tuition state, living with partner in household, parents' university degree, working in the education sector, highest educational attainment, professional degree, employment status, parent status, political party preferences, and voting behavior. Missing values of covariates are imputed; regressions with covariates include imputation dummies. Data source: ifo Education Survey 2015-2020. Robust standard errors clustered at the individual level in parentheses. Significance levels: *** p<0.01, ** p<0.05, * p<0.10.

	Support	for tuition	Opposition a	gainst tuition
	(1)	(2)	(3)	(4)
Income contingency	0.149***	0.146***	-0.150***	-0.145***
	(0.026)	(0.025)	(0.025)	(0.024
Income contingency \times Wave 2016	-0.000	0.002	-0.007	-0.014
	(0.039)	(0.038)	(0.038)	(0.037)
Income contingency × Wave 2018	0.002	0.011	-0.037	-0.046
	(0.041)	(0.041)	(0.040)	(0.039)
Income contingency \times Wave 2019	0.064^{*}	0.068^{**}	-0.062*	-0.070**
	(0.034)	(0.033)	(0.032)	(0.032)
Income contingency \times Wave 2020	0.068^*	0.071**	-0.050	-0.055*
	(0.035)	(0.034)	(0.034)	(0.033)
Wave 2016	-0.014	-0.008	-0.003	0.001
	(0.027)	(0.027)	(0.025)	(0.026)
Wave 2018	0.032	0.021	-0.010	0.004
	(0.029)	(0.029)	(0.027)	(0.004)
Wave 2019	(0.007)	0.001	-0.033	-0.023
	(0.023)	(0.024)	(0.022)	(0.022)
Wave 2020	-0.026	-0.027	0.007	0.013
	(0.025)	(0.025)	(0.023)	(0.023)
Covariates	No	Yes	No	Yes
Observations	12,504	12,502	12,504	12,502

Table A4: Baseline results: Probit model

Notes: Probit regressions, average marginal effects. *Income contingency*: experimental treatment in the survey experiment. Sample: control groups and income-contingency treatment groups in waves 2015, 2016, and 2018-2020. Dependent variable: col. (1)-(2): dummy variables 1 = "strongly favor" or "somewhat favor" tuition, 0 otherwise; col. (3)-(4): dummy variables 1 = "strongly oppose" or "somewhat oppose" tuition, 0 otherwise. Residual category: "neither favor nor oppose." Covariates include age, income, and dummies for gender, born in Germany, living in a tuition state, living with partner in household, parents' university degree, working in the education sector, highest educational attainment, professional degree, employment status, parent status, political party preferences, and voting behavior. Missing values of covariates are imputed; regressions with covariates include imputation dummies. Data source: ifo Education Survey 2015-2020. Regressions weighted by survey weights. Robust standard errors in parentheses. Significance levels: *** p<0.01, ** p<0.05, * p<0.10.

	Support	for tuition	Opposition a	gainst tuition
—	(1)	(2)	(3)	(4)
Income contingency	0.151***	0.151***	-0.177***	-0.173***
	(0.035)	(0.034)	(0.034)	(0.033)
Income contingency × Wave 2016	0.004	0.000	0.007	0.003
	(0.052)	(0.050)	(0.051)	(0.050)
Income contingency × Wave 2018	0.039	0.044	-0.037	-0.039
	(0.063)	(0.062)	(0.061)	(0.060)
Income contingency × Wave 2019	0.045	0.049	-0.019	-0.031
	(0.043)	(0.043)	(0.042)	(0.041)
Income contingency \times Wave 2020	0.070^{*}	0.070^{*}	-0.029	-0.033
	(0.042)	(0.041)	(0.041)	(0.040)
Wave 2016	-0.025	-0.010	-0.014	-0.016
	(0.037)	(0.037)	(0.038)	(0.038)
Wave 2018	0.049	0.049	-0.028	-0.029
	(0.046)	(0.046)	(0.045)	(0.046)
Wave 2019	0.012	0.010	-0.060^{*}	-0.050
	(0.031)	(0.032)	(0.031)	(0.031)
Wave 2020	-0.019	-0.021	-0.015	-0.010
	(0.030)	(0.031)	(0.030)	(0.030)
Covariates	No	Yes	No	Yes
Control mean	0.431	0.431	0.487	0.487
Observations	8,339	8,339	8,339	8,339
R^2	0.038	0.058	0.043	0.067

Table A5: Baseline results: Sample without respondents who participated in more than one survey wave

Notes: OLS regressions. *Income contingency*: experimental treatment in the survey experiment. Sample: control groups and income-contingency treatment groups in waves 2015, 2016, and 2018-2020. Respondents who participated in more than one survey wave are excluded. Dependent variable: col. (1)-(2): dummy variables 1 = "strongly favor" or "somewhat favor" tuition, 0 otherwise; col. (3)-(4): dummy variables 1 = "strongly oppose" or "somewhat oppose" tuition, 0 otherwise. Residual category: "neither favor nor oppose." Control mean: mean of the outcome variable in the control group in wave 2015. Covariates include age, income, and dummies for gender, born in Germany, living in a tuition state, living with partner in household, parents' university degree, working in the education sector, highest educational attainment, professional degree, employment status, political party preferences, and voting behavior. Missing values of covariates are imputed; regressions with covariates include imputation dummies. Data source: ifo Education Survey 2015-2020. Regressions weighted by survey weights. Robust standard errors in parentheses. Significance levels: *** p<0.01, ** p<0.05, * p<0.10.

	Support	for tuition	Opposition a	gainst tuition
—	(1)	(2)	(3)	(4)
Income contingency	0.153***	0.133***	-0.158***	-0.131***
	(0.026)	(0.027)	(0.026)	(0.026)
Income contingency × Wave 2016	-0.000	0.022	-0.007	-0.038
	(0.040)	(0.040)	(0.039)	(0.039)
Income contingency × Wave 2018	0.001	0.031	-0.033	-0.069^{*}
	(0.042)	(0.042)	(0.040)	(0.040)
Income contingency × Wave 2019	0.062^*	0.082^{**}	-0.049	-0.080^{**}
	(0.034)	(0.034)	(0.032)	(0.033)
Income contingency \times Wave 2020	0.068^{*}	0.085^{**}	-0.048	-0.075**
	(0.035)	(0.035)	(0.034)	(0.034)
Wave 2016	-0.014	-0.015	-0.003	0.015
	(0.028)	(0.028)	(0.028)	(0.029)
Wave 2018	0.033	0.013	-0.011	0.017
	(0.029)	(0.030)	(0.029)	(0.029)
Wave 2019	0.007	-0.006	-0.037	-0.014
	(0.024)	(0.025)	(0.024)	(0.024)
Wave 2020	-0.026	-0.033	0.008	0.026
	(0.025)	(0.026)	(0.025)	(0.025)
Covariates	No	Yes	No	Yes
Control mean	0.438	0.438	0.464	0.464
Observations	12,504	12,169	12,504	12,169
R^2	0.036	0.058	0.042	0.064

Table A6: Baseline results: Specification without imputation of missing covariates

Notes: OLS regressions. *Income contingency*: experimental treatment in the survey experiment. Sample: control groups and income-contingency treatment groups in waves 2015, 2016, and 2018-2020. Dependent variable: col. (1)-(2): dummy variables 1 = "strongly favor" or "somewhat favor" tuition, 0 otherwise; col. (3)-(4): dummy variables 1 = "strongly oppose" or "somewhat oppose" tuition, 0 otherwise. Residual category: "neither favor nor oppose." Control mean: mean of the outcome variable in the control group in wave 2015. Covariates include age, income, and dummies for gender, born in Germany, living in a tuition state, living with partner in household, parents' university degree, working in the education sector, highest educational attainment, professional degree, employment status, parent status, political party preferences, and voting behavior. Data source: ifo Education Survey 2015-2020. Regressions weighted by survey weights. Robust standard errors in parentheses. Significance levels: *** p<0.01, ** p<0.05, * p<0.10.

	Strongly favor	Somewhat favor	Neither favor nor oppose	Somewhat oppose	Strongly oppose
	(1)	(2)	(3)	(4)	(5)
Panel A: Adults (Specification of Tab	<i>le 2</i>)				
Income contingency	0.070^{***}	0.079^{***}	0.003	-0.094***	-0.058***
	(0.019)	(0.025)	(0.015)	(0.022)	(0.019)
Income contingency \times Wave 2016	0.016	-0.012	0.010	-0.009	-0.005
	(0.027)	(0.037)	(0.025)	(0.034)	(0.029)
Income contingency \times Wave 2018	-0.020	0.030	0.032	0.005	-0.047^{*}
	(0.032)	(0.040)	(0.023)	(0.036)	(0.028)
Income contingency × Wave 2019	0.044^{*}	0.024	-0.009	-0.037	-0.022
	(0.025)	(0.032)	(0.021)	(0.028)	(0.023)
Income contingency \times Wave 2020	0.025	0.046	-0.017	-0.008	-0.047*
	(0.026)	(0.033)	(0.021)	(0.029)	(0.026)
Wave 2016	-0.003	-0.006	0.008	0.002	-0.002
	(0.018)	(0.026)	(0.018)	(0.025)	(0.021)
Wave 2018	0.048^{**}	-0.027	-0.025*	0.000	0.003
	(0.022)	(0.027)	(0.015)	(0.027)	(0.022)
Wave 2019	0.010	-0.009	0.025	-0.004	-0.022
	(0.016)	(0.023)	(0.015)	(0.021)	(0.017)
Wave 2020	0.027	-0.054**	0.014	-0.017	0.031
	(0.018)	(0.023)	(0.016)	(0.022)	(0.019)
Control mean	0.110	0.329	0.098	0.279	0.185
Observations	12,504	12,504	12,504	12,504	12,504
R^2	0.026	0.023	0.027	0.026	0.043
Panel B: Adolescents (Specification of	f Table 10)				
Income contingency	0.071***	0.066^{**}	0.029	-0.036	-0.130***
<u> </u>	(0.018)	(0.027)	(0.021)	(0.029)	(0.028)
Control mean	0.056	0.211	0.104	0.308	0.321
Observations	1,085	1,085	1,085	1,085	1,085
R^2	0.048	0.036	0.024	0.032	0.062

Table A7: Detailed results by five answer categories

Notes: OLS regressions. Dependent variable: dummy variables 1 = answer category given in respective table header, 0 otherwise. All specifications include covariates. Control mean: mean of the outcome variable in the control group (Panel A: in wave 2015). See the respective tables for additional notes. Data source: ifo Education Survey 2015-2020. Regressions weighted by survey weights. Robust standard errors in parentheses. Significance levels: *** p<0.01, ** p<0.05, * p<0.10.

	Opposing regular tuition but supporting income-contingent tuition					
	Bivariate r	regressions	Multivariat	e regression		
	(1	1)	(2	2)		
Sociodemographic characteristics						
Age	-0.001	(0.001)	-0.001	(0.001)		
Female	0.033	(0.025)	0.026	(0.026)		
Born in Germany	-0.046	(0.058)	-0.036	(0.057)		
Tuition state	0.056^{**}	(0.025)	0.059^{**}	(0.027)		
Monthly household income (1000 €)	0.001	(0.009)	-0.001	(0.011)		
Partner in household	-0.027	(0.026)	-0.029	(0.029)		
Has parent(s) with university degree	0.015	(0.029)	-0.005	(0.033)		
Works in education sector	0.004	(0.050)	-0.008	(0.052)		
Full-time employed	0.011	(0.026)	0.015	(0.035)		
Part-time employed	0.059	(0.042)	0.061	(0.047)		
Self-employed	-0.024	(0.074)	-0.025	(0.076)		
Unemployed	-0.046	(0.043)	-0.031	(0.052)		
Political activity and preferences						
Frequent voter	0.025	(0.056)	0.039	(0.056)		
Education important for vote choice	-0.039	(0.050)	-0.048	(0.050)		
CDU partisan	-0.041	(0.029)	-0.044	(0.030)		
Interest groups						
Middle school degree	-0.012	(0.026)	0.035	(0.030)		
University entrance qualification	0.061^{**}	(0.029)	0.105^{**}	(0.048)		
University degree	0.010	(0.035)	-0.041	(0.050)		
University student	0.040	(0.064)	-0.055	(0.084)		
At least one child < 18	-0.016	(0.030)	-0.011	(0.036)		
All children > 18	0.001	(0.026)	0.038	(0.038)		
Observations	1,5	525	1,5	525		
R^2			0.0)23		

Table A8: Who opposes regular tuition but favors income-contingent tuition?

Notes: OLS regressions. Col. (1): each cell represents the coefficient of a separate OLS regression. Col. (2) depicts one multivariate regression. Sample: control groups in waves 2014 and 2017. Dependent variable: dummy variables 1 = opposing regular tuition but supporting income-contingent tuition, 0 otherwise. Missing values of covariates are imputed; regressions include imputation dummies. Data source: ifo Education Survey 2014, 2017. Regressions weighted by survey weights. Robust standard errors (clustered at the individual level) in parentheses. Significance levels: *** p<0.01, ** p<0.05, * p<0.10.

	Support for income-contingent tuition		Opposition against inc	ome-contingent tuition
	(1)	(2)	(3)	(4)
Earnings information	0.012	0.011	0.004	0.003
	(0.023)	(0.023)	(0.020)	(0.020)
Wave 2017	0.025	0.021	-0.020	-0.017
	(0.030)	(0.029)	(0.025)	(0.025)
Earnings information × Wave 2017	-0.023	-0.020	0.057	0.051
	(0.042)	(0.042)	(0.038)	(0.037)
Covariates	No	Yes	No	Yes
Control mean	0.635	0.635	0.221	0.221
Observations	3,037	3,037	3,037	3,037
R^2	0.000	0.036	0.002	0.040

Table A9: Effects of earnings information on preferences for income-contingent tuition

Notes: OLS regressions. *Earnings information*: experimental treatment in the survey experiment. Sample: waves 2014 and 2017. Dependent variable: col. (1)-(2): dummy variables 1 = "strongly favor" or "somewhat favor" tuition, 0 otherwise; col. (3)-(4): dummy variables 1 = "strongly oppose" or "somewhat oppose" tuition, 0 otherwise. Residual category: "neither favor nor oppose." Control mean: mean of the outcome variable in the control group in wave 2014. Covariates include age, income, and dummies for gender, born in Germany, living in a tuition state, city size, living with partner in household, parents' university degree, working in the education sector, school education, university degree, university student status, employment status, parent status, and political partisanship. Missing values of covariates are imputed; regressions with covariates include imputation dummies. Data source: ifo Education Survey 2014, 2017. Regressions weighted by survey weights. Robust standard errors (clustered at the individual level) in parentheses. Significance levels: *** p<0.01, ** p<0.05, * p<0.10.

	Strongly favor (1)	Somewhat favor (2)	Neither favor nor oppose (3)	Somewhat oppose (4)	Strongly oppose (5)
Income contingency	0.049 [*]	0.112 ^{***}	0.031*	-0.089 ^{***}	-0.103 ^{***}
	(0.026)	(0.030)	(0.017)	(0.027)	(0.019)
Loan framing	0.043*	0.103 ^{***}	0.046 ^{***}	-0.116***	-0.077***
	(0.026)	(0.031)	(0.018)	(0.026)	(0.021)
Covariates	Yes	Yes	Yes	Yes	Yes
Control mean	0.159	0.312	0.075	0.274	0.179
Observations R^2	3,011	3,011	3,011	3,011	3,011
	0.031	0.044	0.049	0.040	0.054

Table A10: The effect of loan framing on preferences for income-contingent tuition: Five answer categories

Notes: OLS regressions. *Income contingency* and *Loan framing*: experimental treatments in the survey experiment. Sample: wave 2018. Dependent variable: dummy variables 1 = answer category given in respective table header, 0 otherwise. Covariates include age, income, and dummies for gender, born in Germany, living in a tuition state, city size, living with partner in household, parents' university degree, working in the education sector, highest educational attainment, professional degree, employment status, parent status, political party preferences, and voting behavior. Missing values of covariates are imputed; regressions with covariates include imputation dummies. Data source: ifo Education Survey 2018. Regressions weighted by survey weights. Robust standard errors in parentheses. Significance levels: *** p<0.01, ** p<0.05, * p<0.10.

	Support for 1	regular tuition	Support for income-contingent tuition			
_	Bivariate regressions	Multivariate regression	Bivariate regressions	Multivariate regression		
	(1)	(2)	(3)	(4)		
Sociodemographic characteristics						
Age	0.001*** (0.000)	0.001* (0.001)	0.002*** (0.000)	0.002*** (0.001)		
Female	-0.071**** (0.013)	-0.066**** (0.014)	-0.027* (0.014)	-0.021 (0.015)		
Born in Germany	0.052* (0.031)	0.026 (0.031)	0.067** (0.032)	0.012 (0.033)		
Tuition state	0.002 (0.014)	-0.002 (0.015)	-0.024 (0.015)	-0.018 (0.015)		
Monthly household income (1000 \in)	0.011** (0.005)	0.006 (0.005)	0.012** (0.005)	0.002 (0.006)		
Partner in household	0.042*** (0.014)	0.017 (0.016)	0.052*** (0.015)	0.021 (0.017)		
Has parent(s) with university degree	-0.001 (0.015)	0.034** (0.016)	-0.019 (0.016)	-0.009 (0.017)		
Works in education sector	0.047** (0.024)	0.072^{***} (0.024)	-0.002 (0.026)	0.022 (0.025)		
Full-time employed	0.059*** (0.014)	0.040** (0.018)	0.034** (0.014)	0.053*** (0.019)		
Part-time employed	-0.017 (0.018)	0.018 (0.021)	-0.023 (0.021)	0.019 (0.023)		
Self-employed	-0.045 (0.033)	-0.027 (0.034)	0.066** (0.031)	0.094*** (0.033)		
Unemployed	-0.072** (0.029)	-0.055* (0.031)	-0.077** (0.031)	-0.012 (0.034)		
Political activity and preferences						
Frequent voter	0.024 (0.019)	0.009 (0.019)	0.079*** (0.021)	0.047** (0.021)		
Education important for vote choice	-0.026 (0.018)	-0.022 (0.018)	0.035* (0.020)	0.034* (0.020)		
CDU partisan	0.090**** (0.016)	0.082*** (0.016)	0.076*** (0.017)	0.062*** (0.017)		
Interest groups						
Middle school degree	0.044^{***} (0.014)	-0.010 (0.017)	0.038** (0.015)	0.019 (0.018)		
University entrance qualification	-0.098*** (0.014)	-0.113*** (0.022)	-0.044**** (0.015)	0.018 (0.023)		
University degree	-0.041** (0.018)	-0.011 (0.022)	-0.034* (0.019)	-0.074*** (0.024)		
University student	-0.188*** (0.033)	-0.090** (0.039)	-0.196**** (0.039)	-0.106** (0.047)		
At least one child < 18	0.018 (0.015)	-0.009 (0.018)	0.012 (0.016)	0.024 (0.020)		
All children > 18	0.006 (0.014)	-0.041** (0.020)	0.071*** (0.015)	0.027 (0.020)		
Observations	9,346	9,346	7,775	7,775		
R^2		0.034		0.036		

Table A11: Who supports tuition?

Notes: OLS regressions. Col. (1) and (3): each cell represents the coefficient of a separate OLS regression. Col. (2) and (4) depict one multivariate regression each. Samples: col. (1)-(2): control groups in waves 2014-2020; col. (3)-(4): income-contingency treatment groups in waves 2014-2020. Dependent variable: dummy variables 1 = "strongly favor" or "somewhat favor" tuition, 0 otherwise. Missing values of covariates are imputed; regressions include imputation dummies. Data source: ifo Education Survey 2014-2020. Regressions weighted by survey weights. Robust standard errors (clustered at the individual level) in parentheses. Significance levels: *** p<0.01, ** p<0.05, * p<0.10.

	Support for tuition		Opposition a	ainst tuition
	(1)	(2)	(3)	(4)
Income contingency	0.142 ^{***} (0.042)	0.133 ^{***} (0.041)	-0.114 ^{***} (0.040)	-0.108*** (0.040)
Income contingency \times Attentive	0.098 ^{**} (0.048)	0.101 ^{**} (0.047)	-0.124*** (0.046)	-0.130*** (0.045)
Attentive	-0.075** (0.035)	-0.074** (0.035)	0.055 (0.035)	0.052 (0.034)
Covariates	No	Yes	No	Yes
Income-contingency effect for attentive respondents	0.239***	0.234***	-0.238***	0.237***
Control mean	0.445	0.445	0.427	0.427
Observations R^2	4,009 0.049	4,009 0.085	4,009 0.052	4,009 0.082

Table A12: The effect of deferred income-contingent payments on preferences for tuition: Heterogeneity by respondents' attention

Notes: OLS regressions. *Income contingency*: experimental treatment in the survey experiment. *Attentive respondent*: dummy variable coded 1 if respondent passed attention check, 0 otherwise. Sample: wave 2019. Dependent variable: col. (1)-(2): dummy variables 1 = "strongly favor" or "somewhat favor" tuition, 0 otherwise; col. (3)-(4): dummy variables 1 = "strongly oppose" or "somewhat oppose" tuition, 0 otherwise. Residual category: "neither favor nor oppose." Significance of income-contingency effects for attentive respondents were calculated using post-estimation Wald tests. Control mean: mean of the outcome variable in the control group. Covariates include age, income, and dummies for gender, born in Germany, living in a tuition state, city size, living with partner in household, parents' university degree, working in the education sector, highest educational attainment, professional degree, employment status, parent status, political party preferences, and voting behavior. Missing values of covariates are imputed; regressions with covariates include imputation dummies. Data source: ifo Education Survey 2019. Regressions weighted by survey weights. Robust standard errors in parentheses. Significance levels: *** p<0.01, ** p<0.05, * p<0.10.

	Support for regular tuition		Support for income	-contingent tuition
	Entered separately	Entered jointly	Entered separately	Entered jointly
	(1)	(2)	(3)	(4)
Despite tuition, poor can afford to study	0.238***	0.125***	0.306***	0.166***
	(0.042)	(0.041)	(0.039)	(0.046)
Tuition is unjust	-0.360***	-0.208***	-0.212***	-0.081
	(0.037)	(0.045)	(0.049)	(0.051)
Tuition makes high earners contribute	0.284***	0.148^{***}	0.331***	0.202^{***}
-	(0.038)	(0.040)	(0.049)	(0.056)
Tuition deters people from studying	-0.250***	-0.071	-0.158***	-0.011
	(0.040)	(0.048)	(0.041)	(0.047)
Tuition restricts choice of subject	-0.242***	-0.047	-0.169***	-0.042
	(0.040)	(0.048)	(0.041)	(0.052)
Tuition increases debt risk	-0.286***	-0.138***	-0.196***	-0.104**
	(0.042)	(0.052)	(0.038)	(0.050)
Tuition is hard to administer	0.056	0.128^{***}	-0.018	-0.038
	(0.048)	(0.041)	(0.042)	(0.041)
Tuition relieves state finance	0.184^{***}	0.080^{**}	0.129***	0.005
	(0.041)	(0.039)	(0.042)	(0.041)
Tuition improves university quality	0.253***	0.128^{***}	0.243***	0.118^{***}
	(0.039)	(0.041)	(0.041)	(0.044)
Covariates	Yes	Yes	Yes	Yes
Control mean	0.4	45	0.6	60
Observations	1,041	1,041	950	950
R^2		0.291		0.243

Table A13: Associations between perceptions about tuition and preferences for tuition

Notes: OLS regressions. Col. (1) and (3): each cell represents the coefficient of a separate regression; col. (2) and (4): one regression each. Samples: col. (1) and (2): subsample of control group asked about indicated statements in wave 2019; col. (3) and (4): subsample of income-contingency treatment group asked about indicated statements in wave 2019. Dependent variable: Dummy variables 1 = "strongly favor" or "somewhat favor" regular/income-contingent tuition, 0 otherwise. Independent variables: dummy variables 1 = "fully applies" or "rather applies", 0 otherwise. Control mean: mean of the outcome variable in the control group. Covariates include age, income, patience, risk tolerance, and dummies for gender, born in Germany, living in a tuition state, city size, living with partner in household, parents' university degree, working in the education sector, highest educational attainment, professional degree, employment status, parent status, political party preferences, and voting behavior. Missing values of covariates are imputed; regressions with covariates include imputation dummies. Data source: ifo Education Survey 2019. Regressions weighted by survey weights. Robust standard errors in parentheses. Significance levels: *** p<0.01, ** p<0.05, * p<0.10.

	Mean in control group	Difference to treatment group			
		Income contingency	Payment deferral	Income threshold	W/o "other countries"
	(1)	(2)	(3)	(4)	(5)
Age	45.72	-1.03	-0.67	-0.99	-0.06
Female	0.50	-0.01	0.01	-0.01	-0.01
Born in Germany	0.94	0.01	-0.01	0.001	0.02
Tuition state	0.72	0.00	-0.01	-0.001	-0.01
City size $\geq 100,000$	0.21	-0.01	-0.02	-0.01	-0.01
Monthly household income (\in)	3,112	-136.5	-51.4	-223.7*	-189.2
Partner in household	0.62	-0.03	0.01	-0.04	-0.03
Parent(s) with univ. degree	0.34	-0.01	-0.04	-0.03	-0.04
Highest education attainment					
No degree/basic degree	0.28	0.01	0.01	-0.00	0.00
Middle school degree	0.31	0.04	0.04	0.05	0.02
University entrance qualif.	0.41	-0.05	-0.05	-0.05	-0.02
Professional degree					
No degree	0.07	0.03^{*}	0.03^{*}	0.04^{**}	0.02
Vocational degree	0.70	-0.02	-0.01	-0.02	0.00
University degree	0.21	-0.03	-0.02	-0.04	-0.03
In training	0.05	0.02	0.00	0.00	0.01
University student	0.02	0.02	0.00	0.00	0.01
Employment status					
Full-time employed	0.50	-0.00	0.05	-0.02	-0.01
Part-time employed	0.16	0.00	0.03	0.00	0.02
Self-employed	0.03	0.00	0.00	0.01	0.02
Unemployed	0.03	0.02	0.01	0.02	0.02
Parent status					
No children	0.46	0.02	-0.02	-0.00	-0.01
At least one child < 18	0.34	-0.04	-0.01	-0.04	-0.03
All children > 18	0.20	0.01	0.03	0.04	0.04
Political party preference					
CDU	0.18	-0.01	-0.03	0.01	0.00
SPD	0.12	0.03	0.02	0.00	0.05^{**}
Linke	0.05	0.03	-0.01	0.01	0.01
Grüne	0.09	-0.03*	0.02	0.02	0.00
Other	0.25	-0.03	0.00	-0.03	-0.04
None	0.32	0.01	-0.01	-0.01	-0.02
Patience (11-point scale)	6.86	-0.32*	-0.05	-0.26	-0.13
Risk tolerance (11-point scale)	5.41	-0.05	-0.03	-0.12	-0.05
Frequent voter	0.79	-0.01	-0.00	-0.02	-0.01
Education important vote	0.78	-0.03	0.01	-0.01	0.02
Non-response	0.00	0.00	0.00	0.00	0.00
Observations	1,095	1,151	1,139	1,137	1,114

Table A14: Summary statistics and balancing tests: Wave 2023

Notes: Weighted group means. "Difference" displays the difference in means between the control group and the treatment groups. Significance levels of "difference" stem from linear regressions of the respective background variable on treatment dummies. Data source: ifo Education Survey 2023. Regressions weighted by survey weights. Significance levels: *** p<0.01, ** p<0.05, * p<0.10.

	Support Neutral Oppose			Total
Regular tuition	(1)	(2)	(3)	(4)
Support	0.352	0.046	0.062	0.460
Neutral	0.061	0.049	0.007	0.117
Oppose	0.212	0.033	0.178	0.423
Total	0.625	0.128	0.247	1.000

Table A15: Within-subject comparison of preferences for regular and income-contingent tuition: Explicit constancy of costs

Notes: Sample: control groups in waves 2014 and 2017. Shares of respondents who support, oppose, or are neutral towards tuition (weighted group means). Data source: ifo Education Survey 2023.

	Support for tuition		Opposition a	gainst tuition	
	(1)	(2)	(3)	(4)	
Income contingency	0.104 ^{***} (0.036)	0.096 ^{***} (0.036)	-0.141*** (0.038)	-0.140 ^{***} (0.037)	
Income contingent \times Direct recruitment	0.119* (0.066)	0.139** (0.066)	-0.083 (0.070)	-0.092 (0.070)	
Direct recruitment	-0.077* (0.043)	-0.091** (0.046)	-0.007 (0.049)	-0.011 (0.052)	
Covariates	No	Yes	No	Yes	
Income-contingency effect on directly recruited respondents	0.223***	0.235***	-0.224***	-0.232***	
Control mean	0.266	0.266	0.629	0.629	
Observations R^2	1,085 0.025	1,085 0.053	1,085 0.031	1,085 0.071	

Table A16: Effects of income-contingent payment schemes on preferences for tuition among adolescents: By recruitment mode

Notes: OLS regressions. *Income contingency*: experimental treatment in the survey experiment. *Direct recruitment*: dummy variable coded 1 if respondent was directly recruited in the online panel, 0 if respondent was recruited via their parents. Sample: adolescent survey, 2018. Dependent variable: col. (1)-(2): dummy variables 1 = "strongly favor" or "somewhat favor" tuition, 0 otherwise; col. (3)-(4): dummy variables 1 = "strongly oppose" or "somewhat oppose" tuition, 0 otherwise. Residual category: "neither favor nor oppose." Control mean: mean of the outcome variable in the control group. Covariates include age, patience, risk tolerance, and dummies for gender, born in Germany, living in a tuition state, city size, living with both parents in household, parents' university degree, expected educational attainment, expected professional degree, parents' employment status, having siblings (who study/studied), and political interest. Missing values of covariates are imputed; regressions with covariates include imputation dummies. Data source: ifo Education Survey 2018 (adolescent survey). Robust standard errors in parentheses. Significance levels: *** p<0.01, ** p<0.05, * p<0.10.

	Control group: Mean	Income contingency: Difference
	(1)	(2)
Age	15.98	0.04
Female	0.47	0.01
Born in Germany	0.98	-0.01
Tuition state	0.73	0.02
City size $\geq 100,000$	0.28	0.01
Lives with both parents	0.74	-0.03
Has parent(s) with university degree	0.38	-0.04
Expected educational attainment		
No degree/basic degree	0.08	-0.01
Middle school degree	0.35	-0.02
University entrance qualification	0.57	0.04
Expected professional degree		
No degree	0.01	-0.01
Vocational degree	0.46	-0.02
University degree	0.52	0.03
Parents' employment status		
Mother full-time employed	0.38	0.01
Mother part-time employed	0.38	-0.01
Mother unemployed	0.04	0.01
Father full-time employed	0.85	-0.05^{*}
Father part-time employed	0.03	0.01
Father unemployed	0.02	0.01
At least one sibling	0.80	-0.04
At least one sibling studies/studied	0.18	-0.04
Politically interested	0.33	0.02
Patience (11-point scale)	6.16	0.25
Risk tolerance (11-point scale)	5.26	-0.03
Observations	525	560

Table A17: Summa	ry statistics and	balancing	tests: Adolescent	t survey
------------------	-------------------	-----------	-------------------	----------

Notes: Sample: adolescent survey, 2018. Weighted group means. "Difference" displays the difference in means between the control group and the treatment group. Significance levels of "difference" stem from linear regressions of the respective background variable on treatment dummies. Data source: ifo Education Survey 2018 (adolescent survey). Significance levels: *** p<0.01, ** p<0.05, * p<0.10.

	Support for regular tuition		Support for incom	Heterogeneous	
	Bivariate regression	Multivariate regression	Bivariate regression	Multivariate regression	treatment effects
	(1)	(2)	(3)	(4)	(5)
Age	-0.007 (0.019)	0.010 (0.020)	0.007 (0.020)	0.013 (0.020)	0.037 (0.061)
Female	-0.025 (0.041)	-0.005 (0.041)	-0.024 (0.045)	-0.006 (0.045)	0.002 (0.061)
Born in Germany	0.075 (0.126)	0.083 (0.129)	0.149 (0.117)	0.129 (0.125)	0.074 (0.172)
Tuition state	-0.012 (0.044)	-0.002 (0.044)	0.034 (0.048)	0.020 (0.048)	0.046 (0.065)
City size $\geq 100,000$	-0.005 (0.043)	-0.013 (0.043)	0.008 (0.046)	0.032 (0.047)	0.013 (0.063)
Lives with both parents	0.048 (0.045)	0.077 (0.047)	-0.005 (0.050)	-0.036 (0.053)	-0.053 (0.067)
Has parent(s) with university degree	0.048 (0.043)	0.097** (0.046)	-0.053 (0.046)	-0.058 (0.048)	-0.101 (0.062)
Expects middle school degree	0.099** (0.045)	0.047 (0.086)	-0.027 (0.048)	-0.029 (0.101)	-0.077 (0.131)
Expects university entrance qualif.	-0.096** (0.043)	-0.021 (0.089)	0.022 (0.046)	-0.030 (0.104)	0.055 (0.124)
Expects university degree	-0.111*** (0.041)	-0.116** (0.055)	0.048 (0.045)	0.044 (0.054)	0.159*** (0.061)
Full-time employed mother	0.038 (0.042)	0.125** (0.053)	-0.085* (0.045)	-0.054 (0.058)	-0.122** (0.062)
Part-time employed mother	0.046 (0.043)	0.122** (0.050)	0.093** (0.046)	0.057 (0.059)	0.047 (0.064)
Full-time employed father	0.043 (0.056)	-0.006 (0.062)	0.084 (0.055)	0.069 (0.067)	0.041 (0.078)
Part-time employed father	-0.087 (0.106)	-0.063 (0.112)	-0.165 (0.101)	-0.081 (0.115)	-0.090* (0.047)
At least one sibling	-0.011 (0.050)	0.003 (0.052)	-0.050 (0.052)	-0.059 (0.053)	-0.039 (0.072)
At least on sibling studies (studied)	-0.077 (0.052)	-0.077 (0.054)	-0.004 (0.063)	0.006 (0.067)	0.072 (0.082)
Politically interested	-0.031 (0.042)	-0.008 (0.046)	0.075 (0.047)	0.064 (0.049)	0.106* (0.063)
Constant		-0.071 (0.355)		0.065 (0.367)	
Observations		525		560	
R^2		0.055		0.036	

Table A18: Correlates of preferences for tuition and heterogeneous treatment effects in the adolescent survey

Notes: Sample: adolescent survey, 2018. OLS regressions. Dependent variable: dummy variables 1 = "strongly favor" or "somewhat favor" tuition, 0 otherwise. Samples: col. (1)-(2): control group; col. (3)-(4): respondents in income-contingency treatment group; col. (5): full adolescent sample. Col. (5) displays coefficients on the interaction term between income-contingency treatment and subgroup indicators from estimates based on equation (2). Missing values of covariates are imputed; regressions include imputation dummies. Data source: ifo Education Survey 2018 (adolescent survey). Regressions weighted by survey weights. Robust standard errors in parentheses. Significance levels: *** p<0.01, ** p<0.05, * p<0.10.